# Motivated Institutional Investors and Firm Investment Efficiency<sup>\*</sup>

Charles Ward<sup>1</sup>, Chao Yin<sup>1</sup>, and Yeqin  $\text{Zeng}^{\dagger 1}$ 

<sup>1</sup>University of Reading

January 13, 2017

#### Abstract

This paper investigates whether firms with higher institutional investors' attention make more efficient investment decisions. Institutions have the most motivation to monitor firms whose holding value are in the top 10% of institutional portfolios. Using the Russell index reconstitution as exogenous shocks to institutional ownership during the 1995-2015 period, we find that U.S firms with high motivated monitoring institutional ownership deviate less from predicted investment levels. The higher motivated institutional ownership is associated with both lower over- and under-investment. Our results are consistent across the different classifications of institution types. Furthermore, we show that the motivated monitoring institutions mitigate over-investment of free cash flow and under-investment due to managers' career concerns. Firms benefit from institution monitoring because our inefficient investment proxies are negatively related to subsequent stock returns. Overall, our results provide new insights into the importance of institution's attention in corporate activities.

Keywords: Institutional investors; Investment efficiency; Monitoring attention; Index switch

JEL classification: G23; G30; G31; M4

<sup>\*</sup>We thank seminar participants at the ICMA Centre, Henley Business School for their insightful and constructive comments. The financial support from ICMA Centre is gratefully acknowledged.

<sup>&</sup>lt;sup>†</sup>Charles Ward. Email: c.ward@icmacentre.ac.uk. ICMA Centre, Henley Business School, University of Reading, Reading, Berkshire RG6 6BA, U.K. Chao Yin. Email: c.yin@icmacentre.ac.uk. ICMA Centre, Henley Business School, University of Reading, Reading, Berkshire RG6 6BA, U.K. Yeqin Zeng. Email: y.zeng@icmacentre.ac.uk. ICMA Centre, Henley Business School, University of Reading, Reading, Berkshire RG6 6BA, U.K.

# 1 Introduction

Institutional investors are important in the capital market because they hold the majority of the market value of U.S stocks (Allen, 2001; Blume and Keim, 2012). The rise of institutional investors over the past four decades (Gompers and Metrick, 2001) positions them as the largest shareholders of most public firms. Compared with retail investors, institutional investors play a more active role in corporate activities. Recent literature uses Russell 1000/2000 index assignments to identify exogenous changes in institutional ownership and documents a positive effect of institutional ownership on corporate activities (e.g. Boone and White, 2015; Appel et al., 2016b; Bird and Karolyi, 2016; Crane et al., 2016; Schmidt and Fahlenbrach, 2016). In practice, it is expensive and time-consuming for investors to collect firm-specific information, analyze the information with professional expertise, monitor firm activities, and intervene through shareholder activism. Given the trade-off between the costs and benefits of active monitoring, institutional investors may not have the same incentive to improve firm governance for the following two reasons. First, institutional investors are heterogeneous at the portfolio level. They differ in terms of types, the length of trading horizons, and activeness (Bushee, 1998; Chen et al., 2007; Schmidt and Fahlenbrach, 2016). Second, the attention of institutional investors is limited. Institutional investors can not evenly monitor all firms held in their portfolios (Kempf et al., 2016) so the motivation of institutional monitoring would depend on the importance of an individual stock in their portfolios (Fich et al., 2015). The previous literature has largely focused on the heterogeneity of institutions and how investors with similar characteristics affect firm performance. Relatively little is known about the divergence of investor monitoring motivation or intensity among all stocks in their portfolios and how this divergence affects managerial decision making.

In this paper, we focus on the limited attention of institutional investors and examine the role of motivated monitoring institutional investors in mitigating corporate inefficient investment. We investigate the monitoring role of institutional investors in investment because investing in projects with positive net present value (NPV) is one of the important determinants of firm future growth. Firms may not invest efficiently due to the conflict of interests between managers and shareholders. Several managerial agency problems leading to over-investment (e.g. Jensen and Meckling, 1976; Shleifer and Vishny, 1997; Richardson, 2006) and under-investment (e.g. Porter, 1991; Bertrand and Mullainathan, 2003; Aghion et al., 2013) are documented. Because higher inefficient investment is associated with lower firm performance (Titman et al., 2004; Cai and Zhang, 2011), understanding whether and how institutional investors improve firm investment decisions by monitoring activities is of particular importance.

When economic agents have limited information-processing capacity, less attention may be rational (Sims, 2003). Based on the assumption of limited attention, Kacperczyk et al. (2016) develop an attention allocation model to predict mutual funds' optimal information choices. Kempf et al. (2016) provide the empirical evidence that an institutional investor's monitoring attention on a certain firm gets distracted if there is an exogenous shock on unrelated firm stocks in its portfolio. Motivated by these studies, we measure an institutional investor' attention to a firm or willingness to monitor the firm by the fraction of the institution's portfolio represented by the firm. We predict that a higher proportion of an institution's portfolio represented by a firm will be associated with higher benefits of monitoring that firm. A firm should get more attention if it is more important to the institution's portfolio. So one can posit that there will be an optimal level of attention determined by this trade-off between monitoring benefits and costs. It is worth noting that the opportunity cost of attention should not be ignored in the trade-off when the attention is in limited supply. An institution's attention on an important firm may get distracted if the institution also monitors unimportant firms in its portfolio. Fich et al. (2015) use this measure as the proxy for the motivation of institutional monitoring and find that targets with more motivated monitoring institutional ownership have higher deal premiums and deal completion probabilities. We extend Fich et al.'s (2015) study to firm investment decisions and also compare the motivated monitoring institutional investors to those with the least motivation to monitor.

Using a large U.S sample for the period of 1995-2015, we measure inefficient investment by the abnormal investment estimated by the investment model developed by Richardson (2006), where over-investment (under-investment) is reflected in a positive (negative) residual. In addition to using Richardson's (2006) single panel regression, we also estimate the inefficient investment in a year by a historical panel regression from 1981 to that year. The historical panel regression method avoids using the unknown firm information in the future to predict the optimal current level of investment. All three measures of inefficient investment are negatively associated with cumulative excess stock returns over the following year, suggesting that institutional investors should pay attention to firm investment decisions. Our measure of institutional investors' motivation to monitor follows Fich et al. (2015). We sort all stocks into ten decile groups by their holding value weights in institutional portfolios. Institutional investors have the highest (least) motivation to monitor firms in the decile 1 (10) groups including the top (bottom) 10% holding rankings. This motivation measure can also be taken as the incentive or intensity of institution monitoring, given that the monitoring attention of institutional investors is limited.

To test the relation between institutional ownership and firm inefficient investment, we must address the endogeneity of institutional ownership. Firms with higher or lower institutional ownership may differ in unobservable characteristics. Therefore, comparing investment efficiency between firms with higher or lower institutional ownership may simply capture the effect of the unobservable differences instead of the effect of institutional investor monitoring. Furthermore, institutional investors may have better information and choose to invest in firms with better investment efficiency and corporate governance. We attempt to mitigate the endogeneity due to the omitted variables and the reverse causality by using the instrumental variables (IV) based on the Russell index annual reconstitution. When firms switch between the Russell 1000 and 2000 indexes, get included in the Russell 2000 index for the first time, or leave the Russell 2000 index, there are exogenous changes in institutional holdings (e.g., Chang et al., 2015; Fich et al., 2015; Crane et al., 2016; Schmidt and Fahlenbrach, 2016). We estimate the relation between institutional ownership and firm inefficient investment in a standard 2-stage least squares (2SLS) estimation framework, similar to Fich et al. (2015) and Schmidt and Fahlenbrach (2016).

Three key findings are yielded in our analyses. First, we show that higher motivated monitoring institutional ownership is associated with the less deviation of investment from its predicted level. Both under- and over-investment are negatively related to motivated monitoring institutional ownership, suggesting that firms which are more closely monitored tend to invest more efficiently. The monitoring role of motivated institutions is economically important. A one standard deviation increase in motivated monitoring institutional ownership results in a \$22.8 million reduction in annual under-investment and a \$60.1 million reduction in annual over-investment for the average sample-size firm with \$2,648.1 million total assets.<sup>1</sup> Second, the effect of the least motivated institutional ownership on firm under-investment is positive and statistically significant, whilst the effect on firm over-investment is insignificant. This result supports our view that the effectiveness of institutional monitoring is influenced by the relative importance of the monitored firms within the institutional portfolios. Third, motivated monitoring investors mitigate the over-investment problem in firms with more cash reserves or free cash flows and mitigate the under-investment problem by reducing the career concerns of firm managers.

Our paper contributes to the literature in four ways. First, we contribute to a growing body of literature that studies the relation between the attention of institutional investors and corporate decision making. Our paper extends Fich et al.'s (2015) study of motivated monitoring institutional investors in M&As to firms' general investment decisions. Kempf et al. (2016) and Liu et al. (2016) find that the monitoring attention of institutional investors on one firm can be distracted by negative shocks to the others in the institutional investors' portfolios. In line with these studies, our results support the limited attention hypothesis that institutional investors can not evenly distribute their monitoring atten-

<sup>&</sup>lt;sup>1</sup>The marginal effect numbers reported here are based on the inefficient investment estimated by the historical panel regressions.

tion among all firms in their portfolios. We show that the effectiveness of institutional monitoring on inefficient investment is reduced as the motivation of monitoring decreases.

Second, our study adds to the studies examining the factors that affect corporate investment, such as free cash flow (Jensen, 1986; Richardson, 2006), earnings management (McNichols and Stubben, 2008), financial reporting quality (Biddle et al., 2009; Cheng et al., 2013; Balakrishnan et al., 2014), management forecast ability (Goodman et al., 2013), product market competition (Gu, 2016; Stoughton et al., 2016), policy uncertainty (Gulen and Ion, 2016), accounting conservatism (Lara et al., 2016), mutual fund flow (Lou and Wang, 2016), and changes in GAAP (Shroff, 2016). Our paper identifies the motivated monitoring institutional ownership as a new factor that reduces firm inefficient investment in both directions.

Third, we shed light on the debate regarding which institutions are more likely to monitor corporate activities. Previous studies find that institutional investors are heterogeneous and only a subset plays an active role in corporate governance (e.g., Brickley et al., 1988; Bushee, 1998; Chen et al., 2007; Cronqvist and Fahlenbrach, 2009; Fich et al., 2015). Recently, Schmidt and Fahlenbrach (2016) find that exogenous increases in passive institutional ownership weaken firm corporate governance and reduce the subsequent firm performance, while Appel et al. (2016b) document that passive mutual funds actually influence firms' governance choices through their large voting blocs and improve firms' longterm performance. We find that all types of motivated monitoring institutional investors do monitor important firms in their portfolios.

Finally, our paper complements a working paper on institutional investors and corporate investment. Wong and Yi (2015) find that firm total institutional ownership is positively related to firm investment and this relation is more pronounced for passive investors than the types of institutional investors. We examine both over- and under-investment in our analyses and use different definitions of inefficient investment. Contrary to Wong and Yi (2015), we document a negative relation between motivated monitoring institutional ownership and firm inefficient investment. More importantly, this negative relation is robust to different types of institutional investors.

The rest of the paper is organized as follows. Section 2 develops our hypotheses. Section 3 describes the details regarding the data sources and variable definitions. Section 4 presents main test results and addresses endogeneity. Section 5 discusses how motivated monitoring institutional investors reduce inefficient investment and provides robustness test results. Finally, Section 6 concludes.

# 2 Hypotheses and empirical predictions

In a perfect capital market without any frictions (Modigliani and Miller, 1958), firm financing and investment decisions are independent. The neoclassical theory of investment predicts that the only determinant of a firm's investment policy is its growth opportunities which are commonly measured by Tobin's Q (Hayashi, 1982; Abel, 1983). Given this ideal framework, the optimal level of investment is achieved when the new investment's marginal benefit is equal to the sum of its marginal cost and the adjustment cost of capital. However, real firm investment usually deviates from the optimal level due to frictions in the capital market such as external financing costs, conflict of interests between managers and shareholders, information asymmetry, managerial optimism or pessimism.

Previous studies suggest that higher inefficient investment is associated with lower subsequent firm performance (e.g. Titman et al., 2004; Cai and Zhang, 2011). Therefore, firm shareholders have a strong incentive to monitor the managers' investment decisions. Institutional investors are usually more active and effective than individual investors in these monitoring activities, because shareholder activism is costly and it is difficult for individual investors to intervene collectively. Nevertheless, the attention of institutional investors is limited and they can not equally monitor all stocks in their portfolio (Kempf et al., 2016). The motivation of institutions to monitor must be positively related to the benefit of monitoring and negatively related to the cost of monitoring. Fich et al. (2015) find that the institutional investors of M&A targets have a greater incentive to monitor when the target stocks are more important relative to their portfolios. Following Fich et al. (2015), we define for each firm, the most motivated (least motivated) monitoring investors as those whose holding value in the firm is in the top (bottom) 10% of their portfolio. Intuitively, the rank of a stock's weight in an institutional investor's portfolio is positively related to the benefit of monitoring. Given limited attention, even if the actual costs of monitoring are equal for all firms in the portfolio, the opportunity cost of monitoring is the highest for the firms in the bottom 10% of the portfolio. When institutional investors monitor the firms in the bottom 10% of their portfolios, attention on the other positions with higher holding value may get distracted. This leads to our first hypothesis:

• *H1:* Motivated monitoring institutional ownership is positively associated with investment efficiency.

Our next two hypotheses examine the inefficient investment direction. Previous studies document two agency problems leading to firm under-investment. First, it takes managers' time and effort to look for positive NPV projects. Managers may instead enjoy "the quiet life" if there is a lack of corporate governance (Hart, 1983; Bertrand and Mullainathan, 2003). Institutional investor monitoring may reduce this kind of managerial slack. Second, the outcomes of new projects remain uncertain, even if managers spend a great amount of effort on supervising the projects. When information is asymmetric, the *ex-ante* quality of new projects and managerial effort are not observable. Many managers' employment and compensation contracts are based on the noisy profitability of projects, instead of the *ex-ante* mean of the projects' returns and actual managerial effort. Managers may not invest on positive NPV projects because the possibility of loss can damage their reputation and job security. Institutional investors may adopt a more professional awareness of the volatility of profitability than other investors. Again et al. (2013) find that institutional investors may reduce managers' career concerns and increase firm innovation activities. Besides these two agency-problem-based explanations, firms may not capture positive NPV investment opportunities due to the debt overhang problem (Hennessy, 2004). Institutional investors may mitigate the debt overhang problem by reducing a firm's debt borrowing cost. The above discussion leads to our second hypothesis:

• H2: Motivated monitoring institutional investors reduce the under-investment problem.

Jensen (1986) predicts that the empire building tendency of managers leads to firm over-investment. The prediction is supported by the empirical findings in Blanchard et al. (1994) that firms over-invest the money from cash windfalls. Harford (1999) finds that firms with higher cash holdings tend to make acquisitions with worse subsequent operation performance, while Richardson (2006) finds that firms with positive free cash flow tend to over-invest. Titman et al. (2004) document a negative relation between over-investment and stock returns, indicating that over-ivestment by managers is not in the interest of shareholders. We expect that a firm with higher motivated monitoring institutional ownership will exhibit less over-investment. Formally stated, our third hypothesis is as follows:

• H3: Motivated monitoring institutional ownership is negatively associated with overinvestment.

# 3 Data and variable description

#### **3.1** Data source

We include U.S firms with available stock returns in the Centre for Research in Security Prices (CRSP) and accounting information in the Compustat Fundamentals Annual files over the 1995–2015 period. Firms in financial (SIC 6000–6999) and regulated utility (SIC 4900–4999) industries are excluded from our sample. Institutional holdings data are obtained from the Thomson Financial CDA/Spectrum Institutional (13F) database. To avoid the potential problems of the reuse of institutional investor identifiers and institution type misclassification in the 13F database, we applied Bushee's data correction to our institutional holdings data. Our sample period is from 1995 to 2015 during which the Russell index constituents data are available on Bloomberg. After these screens, our baseline sample contain 11,903 firms with 92,546 firm–year observations. In addition, we collect the corporate governance measure, *GIndex* scores, from the Institutional Shareholder Services (ISS).

## 3.2 The definition of motivated monitoring institutions

Previous institutional investor literature documents the growth of general institutional investors in the U.S stock market over the past four decades. The two panels of Figure 1 display the time-series plots of institutional investors' market shares and institutional investor numbers at the end of each quarter from 1995 to 2015. Figure 1.1 shows that the percentages of market value held by all institutions, the 100 largest institutions, the 50 largest institutions, and the 10 largest institutions were stable over our sample period. Figure 1.2 presents the sharp increase in the number of institutional investors. The total number of institutions exceeded 3,000 for the first time in the first quarter of 2013. Panel A of Table 1 shows that both the U.S stock market value and the total market value of institutional holdings grew approximately four times over our sample period. Institutional holdings accounted for about 50% of total stock market value in September 1995, the highest level 65.5% in September 2009, and 59.6% in September 2015. The time-series trend of institutional ownership is not a major concern in our study. The annual average number of stocks in an institutional investor's portfolio is over 200, suggesting that a typical institutional investor is unlikely to monitor every firm in its portfolio.

Following Fich et al. (2015), we use stock holding value ranks in an investor's portfolio to differentiate its monitoring motivation among all the stocks in the portfolio. We sort all stocks in an investor's portfolio into ten decile groups by each stock's holding value. As shown in Panel B of Table 1, institutional investors distribute their holding value unevenly across these ten decile groups. On average, more than 40% of their portfolio value is concentrated in the decile 1 group, consisting of the largest 10% of the stocks in their portfolios. Also, the average holding value per stock position (\$105.4 million) in the decile 1 group is almost five times more than it (\$23.7 million) in the decile 2 group. On the other side, only 0.7% of institutional investor portfolio value is in the decile 10 group, consisting of the smallest 10% of the stocks in their portfolios. It is obvious that the performance of the firms in the decile 1 group is much more important than the performance of the rest of firms in institutional investor portfolios. Therefore, the benefit and motivation of monitoring the firms in the decile 1 group is the highest to institutional investors.

Following Fich et al. (2015), we define motivated monitoring institutional investors in a firm as institutional investors whose decile 1 groups include the firm. We aggregate the ownership of all motivated monitoring investors at the firm level and denote the sum as the total motivated monitoring institutional ownership  $Tmi1_{i,t}$ . Similarly, the firms in the decile 10 group are those which institutional investors have the least incentive to monitor. For comparison purposes, we also aggregate the ownership of these institutional investors at the firm level and construct the variable  $Tmil10_{i,t}$ , the ownership of investors who have the least motivation to monitor firms. Panel C of Table 1 shows that on average, the motivated monitoring institutional ownership:  $Nmi1_{i,t}$  the number of motivated monitoring investors only hold 1% of firm shares. We also calculate two alternative measures of motivated monitoring institutional ownership:  $Nmi1_{i,t}$  the number of motivated monitoring institutional investors and  $Pmi1_{i,t}$  the ratio of  $Nmi1_{i,t}$  to the number of firm institutional investors.

### 3.3 Measure of investment inefficiency

We identify inefficient investment as those deviating from the level that would be predicated by a firm-specific model of investment. Motivated by Richardson (2006) and Stoughton et al. (2016), we estimate the following regression and use the residuals as a proxy for the firm-specific inefficient investment:

$$INew_{i,t} = \alpha + \beta_1 \frac{V}{P_{i,t-1}} + \beta_2 Leverage_{i,t-1} + \beta_3 Cash_{i,t-1} + \beta_4 Age_{i,t-1} + \beta_5 Size_{i,t-1} + \beta_6 Return_{i,t-1} + \beta_2 INew_{i,t-1} + \delta_i + \mu_t + \epsilon_{i,t}$$

$$(1)$$

where  $INew_{i,t}$  is the new investment level for firm *i* in year *t* and  $INew_{i,t} = ITotal_{i,t}$  - $IMaintenance_{i,t}$ .  $ITotal_{i,t}$  is the overall investment,  $IMaintenance_{i,t}$  is the investment expenditure to maintain assets in place. Previous finance and economics literature have shown that firm investment is jointly determined by growth opportunities, financial constraints, and other firm characteristics.<sup>2</sup> Firm growth opportunities are measured by V/P, where V is the assets in place and P is the market value of the firm (Ohlson, 1995; Feltham) and Ohlson, 1996).<sup>3</sup> The higher the ratio, the lower the growth opportunities. Therefore we expect a negative relation between V/P and INew. The financial constraints are measured by *Leverage* and *Cash*. Because a lower leverage ratio and higher cash holdings indicate less financial constraints. INew is expected to be negatively related to Leverage and positively related to Cash. Firm characteristics included in equation 1 are firm age (Age), the natural log of firm total assets (Size), cumulative stock returns over the previous year (*Return*), and the lag of new investment. We also include the firm fixed effects  $(\delta_i)$  to control for the unobserved firm characteristics and the year fixed effects  $(\mu_t)$  to control for the factors such as market movement and political cycles.  $\epsilon_{i,t}$  are clustered by firm.<sup>4</sup> Following Richardson (2006), all variables are winsorized at the top and bottom 1%. Please refer to Appendix A for the detailed definitions and construction of these variables.

We define the inefficient investment in our empirical analyses as  $Inef_{i,t} = |INew_{i,t} - \widehat{INew_{i,t}}|$ . As discussed in the previous sections, both under-investment and over-investment are detrimental to the interests of shareholders. But the underlying mechanism of these two problems could be different. We further define the under-investment proxy variable as  $Und_{i,t} = |INew_{i,t} - \widehat{INew_{i,t}}|$  if  $INew_{i,t} < \widehat{INew_{i,t}}$  and the over-investment variable as  $Ovr_{i,t} = |INew_{i,t} - \widehat{INew_{i,t}}|$  if  $INew_{i,t} > \widehat{INew_{i,t}}$ . These two variables help us to distinguish the roles of motivated monitoring institutional investors in mitigating two sources of investment inefficiency.

<sup>&</sup>lt;sup>2</sup>See Hubbard (1998) for a detailed literature review.

<sup>&</sup>lt;sup>3</sup>Richardson (2006) provides the detailed definition of V/P.

<sup>&</sup>lt;sup>4</sup>Petersen (2009) suggests that when the number of firms is much larger than the the number of years, clustering standard errors by firm is similar to double clustering standard errors by firm and year.

To avoid the "look ahead bias" concern due to the use of unknown information at the time of our model prediction, we estimate equation 1 by each year t using the historical data before t, from 1995 to 2015. We trace the sample back to 1981 in order to increase the power of our investment prediction. For example, we run a panel regression from 1981 to 1995 to estimate  $\widehat{INew}_{i,1995}$ , a panel regression from 1981 to 1996 to estimate  $\widehat{INew}_{i,1996}$ , and so on. The predicted investment  $\widehat{INew}_{i,t}$  is the results of 21 historical panel regressions from 1981 to year t, for each t between 1995 and 2015. We denote the inefficient investment proxy variables estimated by this procedure as  $Inef1_{i,t}$ ,  $Und1_{i,t}$ , and  $Ovr1_{i,t}$ .<sup>5</sup> Alternatively, we follow Richardson (2006) and Stoughton et al. (2016) to estimate equation 1 by a single panel regression between 1995 and 2015. The inefficient investment proxy variables are defined as  $Inef2_{i,t}$ ,  $Und2_{i,t}$ , and  $Ovr2_{i,t}$ .

Table 2 reports the corresponding results for the two specifications of equation 1. The left panel displays the average coefficients estimated by the 21 historical panel regressions. The number of negative (-) and positive (+) significant coefficient at the 1% significance level are reported in the parenthesis. The right panel presents the coefficients estimated by the signal panel regression between 1995 and 2015. The negative coefficients of V/P suggest that firms with high growth opportunities increase their investment. The negative (positive) coefficients of *Leverage* (*Cash*) indicate that firms with less financial constraints increase their investment. The negative coefficients of *Size* and *Age* are consistent with the findings in Stoughton et al. (2016) and the firm life cycle hypothesis. The positive coefficients of *Return* and lag *INew* are consistent with Richardson (2006) and Stoughton et al. (2016). The average  $R^2$  of the historical panel and panel regressions are 0.208 and 0.259, suggesting that both of our investment model specifications can explain a significant portion of the variations in firm-specific investment.

<sup>&</sup>lt;sup>5</sup>We also estimate equation 1 with five-year historical rolling windows between year t - 4 and year t. The untabulated results are qualitatively similar.

#### **3.4** Descriptive statistics

Panel C of Table 1 presents the descriptive statistics of all variables in our empirical analyses. The mean and standard deviation of INew are 0.10 and 0.15, which are similar to those (0.08 and 0.13) reported in Richardson (2006). The mean and standard deviation of the difference between INew and  $\widehat{INew}$  estimated by the single panel regression are 0.00 and 0.13, similar to those (0.00 and 0.11) reported in Richardson (2006). The summary statistics of all our investment related variables are also comparable to Stoughton et al. (2016). The mean of our motivated monitoring institutional investor proxies are 0.09 (Tmi1), 9.3 (Nmi1), and 0.03 (Pmi1), which are comparable to those (0.07, 9, and 0.02, correspondingly) reported in Fich et al. (2015). Our proxies are slightly larger because they focus only on institutional investors of M&A targets and their sample spans from 1984 to 2011.

## 4 Main results

In this section, we present the empirical results for the effectiveness of motivated monitoring institutional investors. We first show a negative relation between our inefficient investment proxies and firm financial performance. Then we investigate the role of motivated monitoring institutional investors in firm investment decisions with both ordinary least squares (OLS) and 2SLS regressions. Finally, we study if the monitoring motivation varies across different types of institutional investors.

### 4.1 Inefficient investment and stock returns

Previous empirical studies suggest that inefficient investment has a negative impact on firm performance (Titman et al., 2004). In an efficient market, all information including firm investment decisions will ultimately be transferred into firm stock prices. It is important to show the negative relation between our inefficient investment proxies and subsequent stock returns in order to link institutional monitoring with firm value. To test this relation, we study whether subsequent stock returns of firms with high inefficient investment are significantly lower than those with more efficient investment. If inefficient investment reduces firm value, subsequent stock returns will be lower no matter whether firms have over- or under-invested.

Similar to Daniel and Titman (1997), Faulkender and Wang (2006), and Fich et al. (2016), we measure the subsequent stock returns as the differences between the buy-and-hold returns of our sample firms and the buy-and-hold returns of a benchmark portfolio:

Excess Return<sub>*i*,*t*</sub> = 
$$[\prod_{j=1}^{12} (1 + Ret_{i,j}) - 1] - [\prod_{j=1}^{12} (1 + \text{Benchmark Ret}_{i,j}) - 1]$$
 (2)

where  $Ret_{i,j}$  is the stock return of firm *i* during month *j* and Benchmark Ret<sub>*i*,*j*</sub> is the return of the benchmark portfolio for firm *i* during the same month. Following Fama and French (1993), we sort all the sample firms into quintile portfolios based on their market capitalization on June 30th every year and the book to market ratios at end of December in the previous year. We assign each firm to one of the five-by-five portfolios in every June, and then use the corresponding portfolio as the benchmark portfolio. The mean and median of the excess returns in our sample are -1.6% and -9.1%, which are comparable to those (-0.5% and -8.5%) in Faulkender and Wang (2006).

Next, we regress the excess returns on our inefficient investment proxies:

Excess Return<sub>*i*,*t*+1</sub> = 
$$\alpha + \beta_1$$
Inefficient investment<sub>*i*,*t*</sub> +  $B *$ Control variables<sub>*i*,*t*</sub>  
+  $\theta_i + \epsilon_{i,t}$  (3)

where Inefficient investment is one of the following six variables: Inef1, Und1, Ovr1, Inef2, Und2, and Ovr2. The control variables include MTB, Leverage, Cash, and Size.  $\theta_j$  are the industry fixed effects based on Fama–French 48 industry classification. We estimate equation 3 using a standard panel regression with year fixed effects.  $\epsilon_{i,t}$  are clustered by firm. As an alternative measure, we also use the Fama and Macbeth (1973) regression to estimate equation 3.<sup>6</sup>

The results of these regressions are reported in Table 3. The coefficients of all the inefficient investment proxies are negative and statistically significant. Column (1) of Table 3 suggests that one standard deviation increase in Inef1 is associated with 2.96% decrease in firm annual excess returns. It is noting that the negative effect of inefficient investment on subsequent stock returns is relatively symmetric between under-investment and over-investment. For example, column (3) of Table 3 shows that one standard deviation increase in Und1 is associated with 2.62% decrease in firm annual excess returns, and column (5) of Table 3 shows that one standard deviation increase in Ovr1 is associated with 3.19% decrease in firm annual excess returns.

# 4.2 Institutional ownership and investment inefficiency: Baseline regressions

Given the negative relation between inefficient investment and subsequent stock returns, we expect that strongly motivated monitoring institutional investors will improve firm investment efficiency. To test our three hypotheses, we adapt the following baseline model to capture the effects of institutional investors on investment:

Inefficient investment<sub>*i*,*t*+1</sub> =
$$\alpha + \beta_1 Tmi1_{i,t} + \beta_2 * Tmi10_{i,t} + B * Control variables_{i,t}$$
  
+ $\theta_j + \mu_t + \epsilon_{i,t}$  (4)

where Inefficient investment<sub>*i*,*t*+1</sub> is one of the six proxies: Inef1, Und1, Ovr1, Inef2, Und2, and Ovr2. Tmi1 is the total ownership of motivated monitoring institutions. To compare monitoring roles between the most motivated monitoring investors and the least motivated monitoring investors, we add Tmi10 in equation 4. We test these two variables both individually and jointly. Our control variables are based on Stoughton et al. (2016)

 $<sup>^{6}</sup>$ The panel regression coefficients may be affected more by the years that have more observations. This concern is mitigated by the Fama and Macbeth (1973) regression, in which all years are treated as equally important.

and include *MTB*, *Leverage*, *Cash*, *Size*, *Tangibility*, and *Age*.<sup>7</sup> We also control for the Fama-French 48 industry  $(\theta_j)$  and year  $(\mu_t)$  fixed effects. The standard errors are clustered by firm.

Table 4 presents the results from estimating equation 4. Panel A is based on the inefficient variables estimated by historical panel regressions and Panel B is based on the inefficient variables estimated by the single panel regression. All the coefficients of Tmi1 are negative and statistically significant, suggesting that motivated monitoring institutional investors can enhance investment efficiency. Motivated monitoring institutions also reduce the tendency of firms to under- and over-invest consistently. On the other hand, we find that Tmi10 is positively associated with Inef in both Panel A and Panel B. This positive relation is only statistically significant for the over-investment sample. It is unlikely that the least motivated monitoring investors actually lead to the increase of investment inefficiency. A more rational explanation is that firm managers tend to under-invest when the external monitoring by institutional investors is scarce. Both the quiet life hypothesis and managers' career concern may explain the under-investment tendency of firm managers. We will directly test these two explanations in Section 5.

# 4.3 Institutional ownership and investment inefficiency: Instrumental variable analyses

The results in our baseline regressions may be driven by the endogeneity between motivated monitoring institutional ownership and inefficient investment. The first concern is the potential omitted variable bias. Although we have controlled for several firm characteristics in equation 4, there might be certain unobservable firm characteristics that are correlated with both the motivated monitoring institutional ownership and investment inefficiency. The second concern is the reverse causality that motivated monitoring institutional investors may have inside information on firms' investment efficiency and choose to invest more in those firms with more efficient investment (Giannetti and Simonov, 2006).

<sup>&</sup>lt;sup>7</sup>Appendix A provides the detailed definitions and construction of these variables.

Motivated by the recent studies on firm switching between the Russell 1000 and 2000 indexes, we adopt an IV approach similar to Fich et al. (2015), Schmidt and Fahlenbrach (2016), and Crane et al. (2016).

The Russell 1000 and 2000 indexes are reconstituted in June every year. Based on the market capitalization of firm common stocks as of May 31, the largest 1,000 firms are included in the Russell 1000 index and the subsequent 2,000 firms are included in the Russell 2000 index.<sup>8</sup> In 2005, about \$90 billion institutional assets tracked the Russell 1000 index and above \$200 billion institutional assets tracked the Russell 2000 index (Chang et al., 2015). Both indexes are value-weighted and no other criteria besides the market capitalization is used in the index reconstitution. Therefore, when a stock drops from the Russell 1000 index to the Russell 2000 index or gets newly added in the Russell 2000 index, the institutional ownership of the stock will increase exogenously. On the other hand, there is a negative and exogenous shock on a firm's institutional ownership, when a stock moves up from the Russell 2000 index to the Russell 1000 index or gets excluded from the Russell 2000 index. For the firms which are not affected by the Russell index reconstitution, there is still an exogenous shock on the weights of their holding value in the institutional portfolios when institutional investors adjust their portfolio weights over the annual Russell index reconstitution periods.

The switch of firms between the two Russell indexes and the inclusion/exclusion of firms in the Russell 2000 index are used as the IV in our first stage regression:

$$Tmi1_{i,t} = \alpha + \beta_1 R 1 T R 2_{i,t} + \beta_2 R 2 T R 1_{i,t} + \beta_3 R 2 T N + \beta_4 N T R 2$$
  
+ B \* Control variables\_{i,t} + \theta\_j + \mu\_t + \epsilon\_{i,t} (5)

where  $R1TR2_{i,t}$  ( $R2TR1_{i,t}$ ) is an indicator variable equal to 1 if firm *i* switches from the Russell 1000 (2000) index to the Russell 2000 (1000) index in year *t*.  $R2TN_{i,t}$  ( $NTR2_{i,t}$ ) is an indicator variable equal to one if firm *i* enters (leaves) the Russell 2000 index. The rele-

<sup>&</sup>lt;sup>8</sup>The London Stock Exchange bought Russell Investments in 2014. The merged firm is called FTSE Russell. For the detailed explanations of the Russell Index reconstitution, please refer to http://www.ftserussell.com/research-insights/russell-reconstitution.

vancy condition of our IV is satisfied because the index reconstitution affects the motivated institutional ownership in all firms. The exclusion restriction is also satisfied because the only index assignment rule (the rank of market capitalization) is mechanical and firm stock prices are stochastic. We control for MTB, Leverage, Cash, Size, Tangibility, and Age in equation 5.  $\theta_j$  are Fama–French 48 industry fixed effects and  $\mu_t$  are year fixed effects. In the second step regression, we estimate equation 4 by replacing Tmi1 with  $\widehat{Tmi1}$ , the predicted value of motivated monitoring institutional ownership from equation 5.

Panel A of Table 5 reports the 2SLS regression results. R1TR2, R2TN, and NTR2are negatively associated with Tmi1 in the first stage regression, while R1TR2 is positively correlated with Tmi1. These results are generally in line with Fich et al. (2015). The results of the second stage regressions are presented in columns (1)-(6). In columns (1) and (4), the dependent variables are firm inefficient investment proxies: Inef1 and Inef2. The coefficients of  $\widehat{Tmi1}$  are negative and statistically significant at the 1% level. Our first hypothesis H1 is confirmed. In columns (2) and (5), the dependent variables are under-investment proxies: Und1 and Und2. The coefficients of Tmi1 remain negative and statistically significant at the 1% level. The marginal effect of  $\widehat{Tmi1}$  on the underinvestment proxies is economically significant. For example, a one standard deviation increase in Tmi1 is associated with a 0.86% decrease in Und1 for the average sample-size firm with \$2,648.1 million total assets. This result confirms hypothesis H2. In columns (3) and (6), the dependent variables are over-investment proxies: Ovr1 and Ovr2. The negative and statistically significant coefficients of Tmil confirm that motivated monitoring institutional ownership reduces firm over-investment. The economic significance is such that a one standard deviation increase in Tmi1 results in a 2.27% decrease in Ovr1, which translates into a \$60.1 million reduction in annual over-investment for the average samplesize firm. This result provides direct support for our hypothesis H3.

As an alternative test, we adopt the first difference specification used in Schmidt and Fahlenbrach (2016) and remove any firm-specific time-invariant unobservable firm characteristics. In the 2SLS model estimated in Panel A of Table 5, we replace all the dependent variables and control variables of firm characteristics by their annual change terms. Panel B of Table 5 presents the results of the 2SLS regressions with the first difference specification. All our results remain robust.

## 4.4 Monitoring motivation and institution types

Institutional investors are different in terms of investment strategies, fiduciary duties, and trading horizons. Previous institutional investor studies document that long-term investors and independent investors are more active in firm monitoring than grey investors and short-term investors (Chen et al., 2007). However, Appel et al. (2016b) find that passive mutual funds can actually improve firm governance and long-term performance. Our motivated monitoring investors include all types of investors in the 13F universe. Therefore, a natural question is whether the monitoring motivation of institutional investors varies across different institution types.

First, we divide all motivated monitoring institutional investors into independent  $(Tmi\_Ind)$  and grey investors  $(Tmi\_Grey)$  based on the potential business relationship between institutional investors and their holding firms (e.g., Brickley et al., 1988; Almazan et al., 2005; Chen et al., 2007). Independent investment advisors, investment companies, and public pension funds are classified as independent investors. Private pension funds, banks, and insurance companies are classified as grey investors, because their monitoring ability might be compromised due to business interests.<sup>9</sup> Second, we separate institutional investors into transient  $(Tmi\_Tran)$  and non-transient investors  $(Tmi1\_NonTran)$  based on their investment horizons. Bushee (1998) classifies institutional investors into dedicated, quasi-index, and transient investors based on their investment patterns such as portfolio turnover, diversification, and momentum, etc. Follow Chen et al. (2007), we take transient investors as short-term investors.

Panel A of Table 6 reports the 2SLS regression results of inefficient investment on <sup>9</sup>We follow Bushee's institution type classification for institutional investors after 1998.  $Tmi_Ind$  and  $Tmi_Grey$  and Panel B of Table 6 reports similar regression results for those variables in the first difference terms. Similarly, Panel C of Table 6 reports the 2SLS regression results of inefficient investment on  $Tmi_Tran$  and  $Tmi_NonTran$  and Panel D of Table 6 reports the regression results for those variables in the first difference terms. Among all specifications, the motivated monitoring institutional ownership proxies are negatively related to investment inefficiency. These results suggest that the monitoring incentive derived from the relative importance of firms in the institutional investors' portfolios is independent of investor type characteristics. Even for the institutions which are usually taken as inefficient monitors or passive investors, the benefit of monitoring still increases with the holding firms' relative value in their portfolios. In turn, all motivated monitoring institutional investors improve firm investment efficiency.

## 5 Robustness tests and further discussions

Our results thus far establish a significantly negative relation between motivated monitoring institutional ownership and firm inefficient investment. It is useful to attempt to investigate the channels through which motivated monitoring institutional investors reduce firm over-investment and under-investment. This section concludes with two robustness checks of our main findings.

#### 5.1 Motivated monitoring investors, cash, and over-investment

From an agency perspective, managers have an incentive to over-invest and grow their firms beyond the optimal firm size. This empire building activity increases the resources under the control of firm managers (Jensen, 1986). Previous studies document that the empire building problem is more severe when firms have a substantial amount of free cash flow (e.g., Stulz, 1990; Lang et al., 1991; Brush et al., 2000; Richardson, 2006). Motivated monitoring institutions should have therefore a larger impact on curbing the managers' over-investment tendencies, when firms have more cash reserves or free cash flows. We use following the specification to test this hypothesis:

$$Ovr_{i,t} = \alpha + \beta_1 \widehat{Tmi1}_{i,t} + \beta_2 \widehat{Tmi1}_{i,t} * \text{Cash measures}_{i,t} + \beta_3 \text{Cash measures}_{i,t} + B * \text{Control variables}_{i,t} + \theta_j + \mu_t + \epsilon_{i,t}$$
(6)

where  $\widehat{Tmil}_{i,t}$  is the predicated value of  $Tmil_{i,t}$  in equation 5, Cash measures are either the cash reserve ratio (*Cash*) or the free cash flow (*FCF*), and the other control variables are the same as those in equation 4. We adopt the *FCF* definition in Richardson (2006)<sup>10</sup>:

$$FCF_{i,t} = \text{Operating cash flow}_{i,t} - IMaintaince_{i,t} + R\&D_{i,t} - \widehat{INew}_{i,t}$$
 (7)

Because empire building is usually observed in firms with positive free cash flows (Richardson, 2006), we exclude the negative FCF firm-year observations when we estimate equation 6. The regression results of equation 6, presented in Table 7, show that firms with more cash holdings and free cash flows are more likely to over-invest. This finding is consistent with the prediction that managers may engage in empire building and over-invest when the firms have abundant cash. More importantly, the coefficients ( $\beta_2$ ) of the interaction terms are all negative and statistically significant, indicating that the monitoring role of motivated institutional investors in firm over-investment is more important for firms with the higher empire building tendency.

## 5.2 Quiet life or career concern

The agency problem associated with firm under-investment is that managers do not exert enough effort to seek investment opportunities. There are two possible explanations that predict firm under-investment given the lack of investor monitoring. On the one hand, firm managers overall prefer a quiet life (e.g., Hart, 1983; Bertrand and Mullainathan,

<sup>&</sup>lt;sup>10</sup>Because two different specifications are used to estimate the predicted new investment  $\widehat{INew}$ , we have two measures of free cash flows accordingly.  $\widehat{INew}$  is estimated by the historical panel regressions in FCF1and the single panel regression in FCF2.

2003), because it is costly for firm mangers to seek positive NPV projects and make difficult investment decisions. We call this the "quiet life" or "lazy manager" hypothesis. On the other hand, firm managers are risk averse and do not like to take the risk of new investments. Instead of being lazy, firm managers have job security concerns if the new investment has unfavorable outcomes due to some random factors (Aghion et al., 2013). This possibility is designated as the "career concern" hypothesis.

Although both hypotheses predict that the motivated monitoring investors could reduce firm under-investment, the interaction effect of shareholder monitoring and other external monitoring on under-investment is different. If the "quiet life" hypothesis is correct, monitoring investors have a less important role when market competition is higher. This is because the market competition is positively related to the probability of firm bankruptcy (Hart, 1983). Firm managers are less likely to enjoy "quiet lives" in a competitive market. However, the "career concern" hypothesis predicts the opposite. In a highly competitive market, the probability of new investment failures is larger, which increases the career concerns of firm managers and leads to the under-investment. Monitoring institutional investors may alleviate the managers' career concerns because these investors are informed and can more effectively distinguish the random negative outcomes from the lack of managerial ability. Therefore, monitoring investors have an important role in mitigating under-investment when market competition is greater.

These two hypotheses also have opposite predictions when managerial entrenchment is high. If the "quiet life" hypothesis is correct, managers with the lower risk of being fired have less incentive to seek investment opportunities. Therefore, monitoring investors have a stronger effect on mitigating firm under-investment when managers are more entrenched. If the "career concern" hypothesis is correct, managers have a lower probability of being fired when managerial entrenchment is higher. Therefore, the monitoring investors have a weaker effect on mitigating firm under-investment in this case. We use the following model specification to test these two possible hypotheses:

$$Und_{i,t} = \alpha + \beta_1 \widehat{Tmi1}_{i,t} + \beta_2 \widehat{Tmi1}_{i,t} * \text{Competition or Entrenchment}_{i,t} + \beta_3$$

$$Competition \text{ or Entrenchment}_{i,t} + B * \text{Control variables}_{i,t} + \theta_j + \mu_t + \epsilon_{i,t}$$
(8)

where *Competition* is 1 - Lerner ratio (Aghion et al., 2013) and *Entrenchment* is the governance index *GIndex* (Gompers et al., 2003). The Lerner ratio is the median growth margin of the industry in which firms are assigned.<sup>11</sup> Because the *GIndex* is only available for the S&P 1500 companies between 1995–2007, our sample size in the entrenchment analysis is smaller than it in the main tests.

Table 8 presents the results. Consistent with the "career concern" hypothesis, when market competition is higher and managers are less entrenched, the under-investment problem is more severe. Furthermore, when career concerns are higher, the effect of motivated monitoring investors on the under-investment is more prominent. These results support the "career concern" hypothesis but oppose the "quiet life" hypothesis.

## 5.3 Alternative measures of motivated monitoring investors

In our main analyses, we use the ownership of motivated monitoring institutional investors as our proxy for the strength of monitoring. In this section, we examine two alternative measures of motivated monitoring institutional ownership that have been used in Fich et al. (2015): (1)the proportion of motivated monitoring investors in all the firm institutional investors (Pmi1), and (2) the natural log of one plus the number of motivated monitoring investors (Ln(1 + Nmi1)). We rerun our 2SLS regression analyses, i.e., equations 5 and 4, where we substitute Pmi1 and Ln(1 + Nmi1) as the dependent variables in the first stage regressions and use the predicted value of them in the second stage regressions. The results are tabulated in Panel A and B of Table 9. We find that the negative relation between motivated monitoring institutional ownership and inefficient investment

<sup>&</sup>lt;sup>11</sup>Following Aghion et al. (2013), the Lerner ratio is based on 3-digit SIC codes and the industry fixed effects are based on 4-digit SIC codes in the regressions related to market competition.

is robust to both alternative measures.

# 5.4 Discussions of the identification using the Russell index reconstitution

The identification of our IV is slightly different from those in Fich et al. (2015) and Schmidt and Fahlenbrach (2016). We do not include Russell's float-adjusted market cap-based rankings as IV in our 2SLS regressions. As indicated in Appel et al. (2016a), the float-adjusted market cap-based rankings are affected by insider ownership and liquid outstanding shares. It may overstate the impact of the Russell index reconstitution on the changes in institutional ownership if we include the rankings as IV.<sup>12</sup> The other issue is that Russell adopted a "banding" rule to index assignment in 2007.<sup>13</sup> Although we use the IV method instead of the regression discontinuity method (e.g. Chang et al., 2015) in our paper, we rerun the regressions reported in Table 5 in a restricted sample period 1995-2006. The untabulated test results are similar to those in Table 5.

# 6 Conclusions

Managers may either under-invest or over-invest due to the potential agency problems, which can negatively impact firm subsequent returns. Institutional investors may mitigate firm inefficient investment through monitoring and activism and thereby benefit from the increase in firm value. However, the attention of institutional investors is limited Kempf et al. (2016) and the opportunity cost of monitoring a firm may exceed the benefit if the firm stock's weight is very small in the institutional investor portfolios. Follow Fich et al. (2015), we proxy for the motivation of an institutional investor to monitor a firm by the importance the firm stock in the institutional investor's portfolio. We examine whether motivated monitoring institutional investors can improve firm investment efficiency. By ex-

 $<sup>^{12}</sup>$ Please refer to Appel et al. (2016a) for detailed discussions.

<sup>&</sup>lt;sup>13</sup>Please refer to Crane et al. (2016) for detailed discussions.

tending the measure of abnormal investment developed in Richardson (2006), we show that higher motivated monitoring institutional ownership is associated with both lower underinvestment and over-investment. A similar relation is not found between investors with the least motivation to monitor and inefficient investment. These results are consistent with the limited attention hypothesis.

Our paper contributes to the ongoing debate on whether all types of institutional investors, including the grey and passive ones, contribute to corporate governance. Our evidence strongly suggests that as long as the holdings of a firm stock are important to institutional investors, even grey and passive ones may influence the firm's investment decisions. We also document the channels through which motivated institutional investors mitigate inefficient investment. The role of monitoring investors in reducing over-investment is stronger if firms have higher cash reserves and more free cash flows. Also the role of monitoring investors in reducing under-investment is stronger if firm managers are more likely to have career concern. Overall, our results establish a robust link between institutional investors' motivation and attention and corporate investment decisions.

# Appendix A

### Table A1: Variable Definitions

This table provides variable definitions and corresponding data sources. CRSP refers to the Centre for Research in Security Prices, ISS refers to the Institutional Shareholder Services (formerly RiskMetrics), 13F refers to Thomson Reuters 13F Database, and Bushee's website refers to http://acct.wharton.upenn.edu/faculty/bushee/IIclass.html.

Variable	Definition	Source
Investment reg	ression variables	
AT	Total assets.	Compustat
IT ot al	Annual total investment expenditure normalized by $AT$ :	Compustat
	[Capital expenditure(CAPX) + acquisition	
	expenditure(AQC) + R&D expenditure(XRD)- Receipts from	
	sale of property, plant and equipment (SPPE)]/ $AT$	
	(Richardson, 2006).	
IM a intenance	Annual required investment expenditure to maintain assets in	Compustat
	place normalized by $AT$ :	
	Depreciation and amortization (DPC)/ $AT$ (Richardson, 2006).	
INew	Annual investment expenditure on new projects normalized by	Compustat
	AT: ITotal- IMaintenance (Richardson, 2006).	
MV	Market value of equity: $price(PRCC_F) * common shares$	Compustat
	outstanding (CSHO).	~
V/P	Growth opportunity: Assets in place $/MV$ , where the assets in	Compustat
	place is estimated as $(1 - \alpha r)BV + \alpha (1 + r)X - \alpha rd$ ,	
	$\alpha = \omega/1 + r - \omega$ , $r = 12\%$ , $\omega = 0.62$ , BV is the book value of	
	equity(CEQ), $d$ is annual dividend (DVC), and $X$ is operating	
	income after depreciation (OIADP) (Ohlson, 1995;	
T	Richardson, 2006).	C
Leverage	Leverage ratio: the book value of total debt (long term	Compustat
	debt(DLTT) + short term debt(DLC)) divided by the sum of	
Cash	the book value of total debt and $BV$ (Richardson, 2006).	C
Cash	Cash holding fatio: cash and short term investment(CHE)	Compustat
Ago	divided by A1 at the start of year (Richardson, 2006). Firm are: the natural log of $(1 + \text{the number of years the firm})$	CBSP
nge	has been listed on CBSP as of the start of the year)	0101
	(Richardson, 2006)	
Size	The natural log of $AT$ at the start of the year (Richardson,	Compustat
2020	2006)	compastat
Return	The percent change of firm market value over that prior year:	CRSP
	$MV_t/MV_{t-1} - 1$ (Richardson, 2006).	
MTB	Market to book ratio: market value of asset $(MV+Totaldebt)$	Compustat
	dividend by $AT$ (Stoughton et al., 2016).	

Continued on next page

Variable	Definition	Source
Tangibility	Firm asset tangibility:	Compustat
	Property Plant and Equipment(PPENT) $/AT$ (Stoughton	
	et al., 2016).	
Inef1	Inefficient investment proxy variable: $ INew - \widehat{INew} $ , where	Compustat &
	$\widehat{INew}_t$ is estimated by the historical panel regressions between	CRSP
	1982 and year t.	~
Und1	Under-investment proxy variable: $Inef1$ if $INew < INew$ ,	Compustat &
Ovr1	and zero otherwise. Over-investment proxy variable: $Inef1$ if $INew > \widehat{INew}$ , and	CRSP Compustat &
Inef 2	zero otherwise. Inefficient investment proxy variable: $ INew - \widehat{INew} $ , where	CRSP Compustat &
	$\widehat{INew}$ is estimated by the panel regressions between 1995 and	CRSP
	2015 (Richardson, 2006).	
Und2	Under-investment proxy variable: $Inef2$ if $INew < \widehat{INew}$ ,	Compustat $\&$
	and zero otherwise. $\widehat{I}$	CRSP
Ovr2	Over-investment proxy variable: $Inef2$ if $INew > INew$ , and	Compustat &
	zero otnerwise.	CRSP
Variables relate	ed to institutional investors	105
Nmil	Number of motivated monitoring investors: the number of	13F
	investors whose holding value in the firm is in the top $10\%$ of	
Nmi10	their portfolio (Fich et al., 2015). Number of investors who have the least motivation to monitor	19F
11/11/10	firms: the number of investors whose holding value in the firm	101
	is in the bottom $10\%$ of their portfolio	
Tmi1	Total firm ownership of the motivated monitoring investors	13F
	(Fich et al., 2015).	
Tmi10	Total firm ownership of the investors who have the least	13F
	motivation to monitor firms.	
$Tmi1\_Ind$	Total firm ownership of the motivated monitoring investors	13F & Bushee's
$T_{min}$ : 1 $C_{min}$	who are classified as independent institutional investors.	website
1 mil_Grey	Total firm ownership of the motivated monitoring investors	13F & Bushee's
Tmi1 Tran	who are classified as grey institutional investors. Total firm ownership of the motivated monitoring investors	website 13F & Bushee's
- 11001_1 1 (010	who are classified as transient institutional investors	website
$Tmi1\_NonTran$	Total firm ownership of the motivated monitoring investors	13F & Bushee's
Pmi1	who are classified as non-transient institutional investors. The proportion of motivated institutional investors: the ratio	website 13F
IV is used as a p	of $Nmi1$ to the number of firm institutional investors refix for all predicted institutional investor variables in 2SLS regre	essions.
IV in 2SLS reg	ressions	
R1TR2	Indicator variable: one if a firm switches from the Russell 1000	Bloomberg
	to the Russell 2000 index, and zero otherwise (Fich et al.,	0
	2015).	

Table A1 – continued from previous page

Continued on next page

Variable	Definition	Source
R2TN	Indicator variable: one if a firm drops out of the Russell 2000	Bloomberg
	index due to the decrease of market value, and zero otherwise	
	(Fich et al., 2015).	
R2TR1	Indicator variable: one if a firm switches from the Russell 2000	Bloomberg
	to the Russell 1000 index, and zero otherwise (Fich et al.,	
	2015).	
NTR2	Indicator variable: one if a firm gets included in the Russell	Bloomberg
	2000 index due to the increase of market value, and zero	
	otherwise (Fich et al., 2015).	
Others		
Competition	The industry competition level is defined as 1-Lerner index,	Compustat
	where the Lerner ratio is the industry median gross margin	
	Revenue-Cost of goods sale/ $Revenue$ . Firms are assigned by	
	3-digit SIC codes (Aghion et al., 2013).	
GIndex	The numbers of anti-takeover provisions (Gompers et al.,	ISS
	2003).	
FCF1	Free cash flow: $Operating cash flow(OANCF) -$	Compustat $\&$
	$IMaintenance + R\&D(XRD) - \widehat{INew}$ , where $\widehat{INew}_t$ is	CRSP
	estimated by the histroical panel regressions between 1982 and	
	year $t$ .	
FCF2	Free cash flow: $Operating cash flow(OANCF) -$	Compustat $\&$
	$IMaintenance + R\&D(XRD) - \widehat{INew}$ , where $\widehat{INew}$ is	CRSP
	estimated by the panel regressions between 1995 and 2015 $$	
	(Richardson, 2006).	

Table A1 – continued from previous page

# References

- Abel, A. B., 1983. Optimal investment under uncertainty. The American Economic Review 73, 228–233.
- Aghion, P., Reenen, J. V., Zingales, L., 2013. Innovation and institutional ownership. The American Economic Review 103, 277–304.
- Allen, F., 2001. Do financial institutions matter? Journal of Finance 4, 1165–1175.
- Almazan, A., Hartzell, J. C., Starks, L. T., 2005. Active institutional shareholders and costs of monitoring: Evidence from executive compensation. Financial Management 34, 5–34.
- Appel, I., Gormley, T. A., Keim, D. B., 2016a. Identification using russell 1000/2000 index assignments: A discussion of methodologies. Working Paper, Available at SSRN 2641548.
- Appel, I. R., Gormley, T. A., Keim, D. B., 2016b. Passive investors, not passive owners. Journal of Financial Economics 121, 111–141.
- Balakrishnan, K., Core, J. E., Verdi, R. S., 2014. The relation between reporting quality and financing and investment: Evidence from changes in financing capacity. Journal of Accounting Research 52, 1–36.
- Bertrand, M., Mullainathan, S., 2003. Enjoying the quiet life? Corporate governance and managerial preferences. Journal of Political Economy 111, 1043–1075.
- Biddle, G. C., Hilary, G., Verdi, R. S., 2009. How does financial reporting quality relate to investment efficiency? Journal of Accounting and Economics 48, 112–131.
- Bird, A., Karolyi, S. A., 2016. Do institutional investors demand public disclosure? Review of Financial Studies 29, 3245–3277.
- Blanchard, O. J., Lopez-de Silanes, F., Shleifer, A., 1994. What do firms do with cash windfalls? Journal of Financial Economics 36, 337–360.
- Blume, M. E., Keim, D. B., 2012. Institutional investors and stock market liquidity: trends and relationships. Working Paper, The Wharton School, University of Pennsylvania, Philadelphia, PA.
- Boone, A. L., White, J. T., 2015. The effect of institutional ownership on firm transparency and information production. Journal of Financial Economics 117, 508–533.
- Brickley, J. A., Lease, R. C., Jr., C. W. S., 1988. Ownership structure and voting on antitakeover amendments. Journal of Financial Economics 20, 267–291.
- Brush, T. H., Bromiley, P., Hendrickx, M., et al., 2000. The free cash flow hypothesis for sales growth and firm performance. Strategic Management Journal 21, 455–472.
- Bushee, B. J., 1998. The influence of institutional investos on myopic R&D investment behavior. The Accounting Review 73, 305–333.

- Cai, J., Zhang, Z., 2011. Leverage change, debt overhang, and stock prices. Journal of Corporate Finance 17, 391–402.
- Chang, Y.-C., Hong, H., Liskovich, I., 2015. Regression discontinuity and the price effects of stock market indexing. Review of Financial Studies 28, 212–246.
- Chen, X., Harford, J., Li, K., 2007. Monitoring: Which institutions matter? Journal of Financial Economics 86, 279–305.
- Cheng, M., Dhaliwal, D., Zhang, Y., 2013. Does investment efficiency improve after the disclosure of material weaknesses in internal control over financial reporting? Journal of Accounting and Economics 56, 1–18.
- Crane, A. D., Michenaud, S., Weston, J. P., 2016. The effect of institutional ownership on payout policy: Evidence from index thresholds. Review of Financial Studies hhw012, 1–32.
- Cronqvist, H., Fahlenbrach, R., 2009. Large shareholders and corporate policies. Review of Financial Studies 22, 3941–3976.
- Daniel, K., Titman, S., 1997. Evidence on the characteristics of cross sectional variation in stock returns. Journal of Finance 52, 1–33.
- Fama, E. F., French, K. R., 1993. Common risk factors in the returns on stocks and bonds. Journal of Financial Economics 33, 3–56.
- Fama, E. F., Macbeth, J., 1973. Risk, return, and equilibrium: Empirical tests. Journal of Political Economy 70, 607–636.
- Faulkender, M., Wang, R., 2006. Corporate financial policy and the value of cash. Journal of Finance 61, 1957–1990.
- Feltham, G. A., Ohlson, J. A., 1996. Uncertainty resolution and the theory of depreciation measurement. Journal of Accounting Research 34, 209–234.
- Fich, E. M., Harford, J., Tran, A. L., 2015. Motivated monitors: The importance of institutional investors' portfolio weights. Journal of Financial Economics 118, 21–48.
- Fich, E. M., Harford, J., Yore, A. S., 2016. The effect of takeover protection on the value of cash: Evidence from a natural experiment. Available at SSRN 2748086.
- Giannetti, M., Simonov, A., 2006. Which investors fear expropriation? Evidence from investors' portfolio choices. Journal of Finance 61, 1507–1547.
- Gompers, P. A., Ishii, J. L., Metrick, A., 2003. Corporate governance and equity prices. Tech. rep.
- Gompers, P. A., Metrick, A., 2001. Institutional investors and equity prices. Quarterly Journal of Economics 116, 229–259.
- Goodman, T. H., Neamtiu, M., Shroff, N., White, H. D., 2013. Management forecast quality and capital investment decisions. The Accounting Review 89, 331–365.

- Gu, L., 2016. Product market competition, R&D investment, and stock returns. Journal of Financial Economics 119, 441–455.
- Gulen, H., Ion, M., 2016. Policy uncertainty and corporate investment. Review of Financial Studies 29, 523–564.
- Harford, J., 1999. Corporate cash reserves and acquisitions. Journal of Finance 54, 1969–1997.
- Hart, O. D., 1983. The market mechanism as an incentive scheme. The Bell Journal of Economics 14, 366–382.
- Hayashi, F., 1982. Tobin's marginal q and average q: A neoclassical interpretation. Econometrica 50, 213–224.
- Hennessy, C. A., 2004. Tobin's q, debt overhang, and investment. Journal of Finance 59, 1717–1742.
- Hubbard, R. G., 1998. Capital-market imperfections and investment. Journal of Economic Literature 36, 193–225.
- Jensen, M. C., 1986. Agency costs of free cash flow, corporate finance, and takeovers. The American Economic Review 76, 323–329.
- Jensen, M. C., Meckling, W. H., 1976. Theory of the firm: Managerial behavior, agency costs and ownership structure. Journal of Financial Economics 3, 305–360.
- Kacperczyk, M., Van Nieuwerburgh, S., Veldkamp, L., 2016. A rational theory of mutual funds' attention allocation. Econometrica 84, 571–626.
- Kempf, E., Manconi, A., Spalt, O. G., 2016. Distracted shareholders and corporate actions. Review of AFinancial Studies, Forthcoming.
- Lang, L. H., Stulz, R. M., Walkling, R. A., 1991. A test of the free cash flow hypothesis: The case of bidder returns. Journal of Financial Economics 29, 315–335.
- Lara, G., Juan, M., Garcia Osma, B., Penalva, F., 2016. Accounting conservatism and firm investment efficiency. Journal of Accounting and Economics 61, 221–238.
- Liu, C. Y., Low, A., Masulis, R., Zhang, L., 2016. Monitoring the monitor: Distracted institutional investors and board governance. Working Paper.
- Lou, X., Wang, A. Y., 2016. Flow-induced mispricing and corporate investment. Working Paper, Available at SSRN 2169135.
- McNichols, M. F., Stubben, S. R., 2008. Does earnings management affect firms' investment decisions? The Accounting Review 83, 1571–1603.
- Modigliani, F., Miller, M. H., 1958. The cost of capital, corporation finance and the theory of investment. The American Economic Review 48, 261–297.

- Ohlson, J. A., 1995. Earnings, book values, and dividends in equity valuation. Contemporary accounting research 11, 661–687.
- Petersen, M., 2009. Estimating standard errors in finance panel data sets: Comparing approaches. Review of Financial Studies 22, 435–480.
- Porter, M. E., 1991. Capital disadvantage: America's failing capital investment system. Harvard Business Review 70, 65–82.
- Richardson, S., 2006. Over-investment of free cash flow. Review of accounting studies 11, 159–189.
- Schmidt, C., Fahlenbrach, R., 2016. Do exogenous changes in passive institutional ownership affect corporate governance and firm value? Journal of Financial Economics Forthcoming.
- Shleifer, A., Vishny, R. W., 1997. A survey of corporate governance. Journal of Finance 52, 737–783.
- Shroff, N., 2016. Corporate investment and changes in GAAP. Review of Accounting Studies, Forthcoming.
- Sims, C. A., 2003. Implications of rational inattention 50, 665–690.
- Stoughton, N., Wong, K. P., Yi, L., 2016. Investment efficiency and product market competition. Journal of Financial and Quantitative Analysis, Forthcoming.
- Stulz, R., 1990. Managerial discretion and optimal financing policies. Journal of Financial Economics 26, 3–27.
- Titman, S., Wei, K. J., Xie, F., 2004. Capital investments and stock returns. Journal of financial and Quantitative Analysis 39, 677–700.
- Wong, K. P., Yi, L., 2015. Institutional investors and corporate investment. Working Paper.

Figure 1: Institutional ownership over the sample period 1995–2015

Figure 1.1. Market shares of institutional investors. This figure presents the cumulative institutional holdings as a percent of total U.S. stock market value for all, top 10, top 50, and top 100 institutional investors, respectively. The sample period is from March 1995 to December 2015.



Figure 1.2. Number of institutional investors. This figure plots the number of institutional investors in the U.S. stock market. The sample period is from March 1995 to December 2015.



statistics
Summary
÷
Table

of stocks on the market, the number of institutional investors, the total market value of institutional holdings (billion dollars), the stock market value (billion dollars), the percentage of stock market share held by institutions, and the average number of stocks in Panel A. Time series of institutional holdings. This panel presents the number of reported institutional positions, the number institutional investor portfolios. We report these summary statistics in every September from 1995 to 2015.

Time	Reported positions	Number of stocks	Number of institutions	Market value of institutional holdings	Stock market value	Percentage	Average number of stocks per institution
Sep-95	343,187	8,256	1,212	3,303	6,570	50.0%	283
Sep-96	340,827	8,760	1,174	3,896	7,951	49.6%	290
Sep-97	400,135	9,062	1,363	5,677	10,898	52.3%	294
Sep-98	428,053	9,170	1,484	5,761	11,232	53.3%	288
Sep-99	459, 388	8,871	1,553	8,260	14,232	50.9%	296
Sep-00	514,160	9,071	1,740	9,699	18,102	54.3%	295
Sep-01	491,891	8,604	1,822	6,997	12,615	55.4%	270
Sep-02	504,951	8,175	1,878	6,061	10,495	56.7%	269
Sep-03	524,618	7,432	1,867	7,843	13,283	54.0%	281
Sep-04	574, 246	7,733	2,045	9,443	15,343	61.2%	281
Sep-05	605,990	7,718	2,224	11,002	17,694	64.5%	272
Sep-06	654, 812	7,607	2,447	12,333	19,113	64.8%	268
Sep-07	690,667	7,618	2,656	15,036	22,036	62.9%	260
Sep-08	683,039	7,503	2,834	10,966	16,680	64.7%	241
Sep-09	660, 354	7,114	2,702	10,154	15,538	65.5%	244
Sep-10	664, 732	7,009	2,705	11,101	17,092	65.2%	246
Sep-11	673, 870	6,549	2,895	10,779	16,568	64.6%	233
Sep-12	673, 732	6,139	2,893	12,769	20,517	63.7%	233
Sep-13	687, 379	5,893	3,018	13,524	23,669	62.2%	228
Sep-14	714,804	5,849	2,974	14,910	26,842	54.7%	240
Sep-15	649, 619	5,231	2,732	13,686	23,274	59.6%	238

	Individual stock	Deci	le portfoli	io value to total p	oortfolio value
$\operatorname{Group}$	Average holding value	Mean	Median	25th percentile	75th percentile
Decile 1	105,443.7	41.5%	38.1%	27.2%	53.0%
Decile 2	23,676.6	18.8%	18.9%	15.9%	21.7%
Decile 3	11,977.7	12.5%	13.1%	9.9%	15.3%
Decile 4	6,999.6	8.6%	9.0%	5.9%	11.3%
Decile 5	4, 339.2	6.1%	6.1%	3.5%	8.6%
Decile 6	2,790.4	4.7%	4.3%	2.2%	6.8%
Decile 7	1,745.8	3.3%	2.8%	1.3%	4.7%
Decile 8	1,057.9	2.3%	1.8%	0.8%	3.2%
Decile 9	594.2	1.5%	1.1%	0.5%	2.1%
Decile 10	235.7	0.7%	0.5%	0.2%	1.0%

**Panel C. Main variables.** This panel presents the descriptive statistics of firm and institutional investor variables in our sample. The sample period is between 1995 and 2015. All firms have complete information in the CRSP and Compustat databases. We also require that our sample firms have institutional ownership information from the Thomson Financial CDA/Spectrum Institutional (13F) database. The number of observations, mean, standard deviation, minimum, 25th percentile, median, 75th percentile, and maximum are reported from left to right in sequence for each variables. Detailed definitions of all variables can be found in Appendix A.

Variable	Obs.	Mean	S.D.	Min	$\mathbf{p25}$	Median	$\mathbf{p75}$	Max
Investment regress	ion varia	bles						
AT	$92,\!546$	$2,\!648.1$	7,563.3	2.3	62.2	265.5	1,302.2	$47,\!604.0$
ITotal	$92,\!546$	0.15	0.15	-0.03	0.05	0.10	0.20	0.74
IM a intenance	$92,\!546$	0.05	0.04	0.00	0.03	0.04	0.06	0.24
INew	$92,\!546$	0.10	0.15	-0.18	0.00	0.05	0.14	0.71
MV	$92,\!546$	3,864.9	$17,\!935.9$	0.0	60.7	281.6	1,327.0	630,000.0
V/P	$92,\!546$	0.49	0.71	-2.74	0.20	0.44	0.74	3.07
Leverage	$92,\!546$	0.31	0.33	0.00	0.01	0.24	0.48	1.71
Cash	$92,\!546$	0.22	0.30	0.00	0.02	0.10	0.30	1.59
Age	$92,\!546$	2.24	1.01	0.00	1.61	2.30	3.00	4.19
Size	$92,\!546$	5.35	2.46	0.00	3.79	5.36	7.01	10.69
Return	$92,\!546$	0.21	0.85	-0.87	-0.25	0.00	0.39	4.48
MTB	$92,\!546$	1.86	1.81	0.27	0.83	1.25	2.10	11.10
Tangibility	$92,\!546$	0.26	0.24	0.00	0.08	0.18	0.39	0.91
Excess Return_ $FF25$	$78,\!602$	-0.02	0.57	-2.03	-0.33	-0.09	0.18	14.63
$INew - \widehat{INew}_1$	84,731	0.01	0.13	-0.43	-0.06	-0.01	0.05	0.77
$INew - \widehat{INew}_2$	84,731	0.00	0.13	-0.44	-0.07	-0.02	0.04	0.76
Inef1	84,731	0.09	0.10	0.00	0.03	0.06	0.11	0.77
Und1	$47,\!613$	0.07	0.05	0.00	0.03	0.06	0.09	0.43
Ovr1	$37,\!118$	0.11	0.13	0.00	0.03	0.06	0.15	0.77
Inef2	84,731	0.09	0.09	0.00	0.03	0.06	0.11	0.76
Und2	$47,\!613$	0.08	0.05	0.00	0.04	0.07	0.10	0.44
Ovr2	$37,\!118$	0.10	0.13	0.00	0.02	0.05	0.13	0.76
Variables related to	o institut	tional inv	vestors					
Tmi1	$92,\!546$	0.09	0.15	0.00	0.00	0.00	0.13	1.00
Tmi10	$92,\!546$	0.01	0.02	0.00	0.00	0.00	0.01	1.00
Nmi1	$92,\!546$	9.3	41	0	0	0	3	1,058
Nmi10	$92,\!546$	9.2	10	0	3	6	12	295
$Tmi1\_Ind$	$92,\!546$	0.07	0.11	0.00	0.00	0.00	0.10	1.00
$Tmi1\_Grey$	$92,\!546$	0.02	0.06	0.00	0.00	0.00	0.00	1.00
$Tmi1\_Tran$	$92,\!546$	0.02	0.05	0.00	0.00	0.00	0.03	1.00
$Tmi1\_NonTran$	$92,\!546$	0.06	0.12	0.00	0.00	0.00	0.08	1.00
Pmi1	$92,\!546$	0.03	0.06	0.00	0.00	0.00	0.04	1.00
Instrumental varial	bles in 2	SLS regr	essions					
R1TR2	$94,\!648$	0.01	0.10	0.00	0.00	0.00	0.00	1.00
R2TN	$94,\!648$	0.03	0.17	0.00	0.00	0.00	0.00	1.00
R2TR1	$94,\!648$	0.01	0.11	0.00	0.00	0.00	0.00	1.00
NTR2	$94,\!648$	0.04	0.19	0.00	0.00	0.00	0.00	1.00
Others								
Competition	$92,\!545$	0.63	0.15	0.15	0.53	0.64	0.74	3.04
GIndex	$7,\!317$	8.94	2.66	1.00	7.00	9.00	11.00	17.00
FCF1	84,731	-0.04	0.18	-0.92	-0.10	-0.01	0.06	0.60
FCF2	84,731	-0.05	0.18	-0.94	-0.11	-0.02	0.05	0.50

#### Table 2: Optimal investment expenditure regressions

This table reports the regression coefficients of the optimal investment expenditure model developed by Richardson (2006). The dependent variable is *INew* measured in year t. The independent variables are V/P, *Leverage*, *Cash*, *Size*, *Return*, *Age*, and *INew* measured in year t - 1. Detailed definitions of these variables are described in Appendix A. In the historical panel regressions, we run a panel regression with firm year observations between 1982 and year t, for each year t of our sample period 1995-2015. We only report the time series average of the coefficients estimated by twenty-one historical panel regressions. The numbers of positive and negative coefficients, with 1% significance level, are reported in parenthesis. In the panel regressions, we run one panel regression over our sample period 1995-2015. The standard errors are clustered by firm in both regressions. t-values are reported in brackets. \*\*\*, \*\*, and \* denote statistical significance at the 1%, 5%, and 10% levels, respectively. Firm and year fixed effects are controlled in all regressions.

Historical panel regre	essions	Panel regres	sions
$\frac{V}{P}_{t-1}$	-0.016	$\left  \frac{V}{P}_{t-1} \right $	-0.021***
	(-21, +0)	101	[-19.26]
$Leverage_{t-1}$	-0.091	$Leverage_{t-1}$	-0.082***
	(-21, +0)		[-24.76]
$Cash_{t-1}$	0.033	$Cash_{t-1}$	$0.030^{***}$
	(-0, +21)		[10.10]
$Size_{t-1}$	-0.008	$Size_{t-1}$	-0.009***
	(-21, +0)		[-18.42]
$Return_{t-1}$	0.009	$Return_{t-1}$	$0.007^{***}$
	(-0, +21)		[11.35]
$Age_{t-1}$	-0.011	$Age_{t-1}$	-0.002
	(-21, +0)		[-1.01]
$INew_{t-1}$	0.115	$INew_{t-1}$	$0.124^{***}$
	(-0, +21)		[19.24]
Constant	0.166	Constant	$0.166^{***}$
	(-0, +21)		[50.69]
Average Observation	89,129	Observations	84,731
Average Adj. R-Squared	0.208	Adj. R-squared	0.259
Year fixed effects	Yes	Year fixed effects	Yes
Firm fixed effects	Yes	Firm fixed effects	Yes
Number of historical panels	21		

mulative stock returns in year $t + 1$ on the investment inefficiency proxy variables	, and the Fama and Macbeth (1973) (F&M) regressions are reported. Our sample	ie dependent variable is the firm cumulative annual stock return and the independent	ted by the investment regressions reported in Table 2. $Inef1$ , $Und1$ , and $Ovr1$ are	nd2, and $Ovr2$ are estimated by the panel regressions. Detailed definitions of the	na-French 48 industry fixed effects are controlled for in all regressions. Year fixed	ed by firm in the panel regressions. t-values are reported in brackets. ***, **, and
regression results of the firm cumulative stock returns in	ints of both the panel regressions and the Fama and Mac	vears for the period $1995-2015$ . The dependent variable is the	the inefficient investment estimated by the investment re-	rical panel regressions. $Inef2$ , $Und2$ , and $Ovr2$ are estim	are described in Appendix A. Fama–French 48 industry fi	nd the standard errors are clustered by firm in the panel re
This table presents th	in year $t$ . The coeffici	consists of 68, 840 firm	variables of interest ar	estimated by the histo	independent variables	effects are controlled a

Variables	(1) Panel	(2) F&M	(3) Panel	$^{(4)}_{ m F\&M}$	(5) Panel	(6) F&M	(7) Panel	$^{(8)}_{ m F\&M}$	(9) Panel	(10) F&M	(11) Panel	(12) F&M
Inef1	$-0.296^{***}$	$-0.264^{***}$										
Und1	[/0.0T-]	[11:0-]	$-0.524^{***}$	$-0.310^{***}$								
Ovr1			[70.7]	-3.29]	$-0.245^{***}$	-0.237***						
Inef2					[66.7-]	[17.7-]	$-0.297^{***}$	$-0.264^{***}$				
Und2							[-10.22]	[-8.42]	$-0.552^{***}$	$-0.346^{***}$		
Ovr2									[-7.73]	[-3.73]	$-0.249^{***}$	-0.235***
MTB	-0.003	-0.006*	$-0.010^{***}$	$-0.012^{***}$	0.007**	0.003	-0.003	-0.006**	$-0.010^{***}$	$-0.012^{***}$	[cc.7-]	0.003
Leverage	$[-1.16]$ $0.035^{***}$	[-2.09] 0.026	[-3.36] -0.022	[-4.22] -0.023	$[2.04] 0.119^{***}$	[0.78] 0.105***	[-1.27] 0.033***	[-2.13] 0.025	[-3.34] -0.020	[-4.22]-0.021	$[2.02] 0.117^{***}$	$[0.73] 0.103^{***}$
Cash	[3.20]-0.069***	[0.98] -0.036	[-1.55] -0.053***	[-0.85]	[6.86] -0.078***	[3.39] -0.047	[3.04]-0.070***	[0.93]-0.036	[-1.36]-0.057***	[-0.77]	[6.77] -0.077***	[3.35] $-0.047$
0	[-5.78]	[-1.10]	[-3.33]	[-0.53]	[-4.25]	[-1.44]	[-5.88]	[-1.12]	[-3.53]	[-0.59]	[-4.22]	[-1.42]
əzic	[-21.49]	-0.024 [-7.99]	-0.030 [-16.98]	-0.020	-0.022	-0.020 [-5.22]	[-21.59]	-0.024 [-8.06]	-0.030	-0.02/	[-11.83]	-0.020
Constant	$0.109^{**}$	$0.291^{***}$	$0.169^{***}$	$0.337^{***}$	0.021 0.61	0.069 [1 27]	$0.113^{***}$	$0.294^{***}$	$0.182^{***}$	$0.341^{***}$	0.019 0.541	0.068
	[00.±]	07.7	[10.0]	0.04		[ J 7 · T	[±0.0]	61.1	[T7:0]	[00.0]	[±0.0]	[12.1]
Observations	68,840	68,840	38,829	38,829	30,011	30,011	68,840	68,840	38,829	38,829	30,011	30,011
Adj. R-squared	0.016	0.092	0.016	0.092	0.018	0.120	0.016	0.091	0.016	0.092	0.018	0.120
Industry fixed effects Vear fixed effects	${ m Y}_{ m es}$	Yes	${ m Y}_{ m es}$	Yes	Yes Ves	Yes	Yes Yes	Yes	Yes Yes	Yes	Yes Yes	Yes
Number of groups	5	21	3	21	3	21	3	21	3	21	3	21

regressions
Baseline
:: :
investmen
Ŀ
inefficien
l ownership an
al
institution
Ч
Motivate
<u></u>
J J
abl€

of the firm inefficient investment on the institutional ownership. Our sample consists of 80,031 firm years for the period 1995–2015. The effects are controlled for in all regressions. The standard errors are clustered by firm. t-values are reported in brackets. \*\*\*, \*\*, and \* The independent variables of interest are the most motivated institutional investor ownership (Tmi1) and the least motivated institutional investor ownership (Tmi10). Detailed definitions of all variables are described in Appendix A. Fama-French 48 industry and year fixed Panel A. Inefficient investment estimated by the historical panel regressions. This panel presents the panel regression results dependent variables are the firm inefficient investment proxy variables: Inef1, Und1, and Ovr1 estimated by the historical panel regressions. denote statistical significance at the 1%, 5%, and 10% levels, respectively.

Variables         (1) $Tmi1$ $-0.007^{**}$ $Tmi10$ $[-2.26]$ $Tmi10$ $[-3.26]$ $MTB$ $[-3.26]$ $MTB$ $[-3.26]$ $Leverage$ $0.006^{***}$ $Leverage$ $0.008^{***}$	(2) 0.079*** [3.57] 0.006*** [14.17] 0.008***	(3) -0.006*	(4)	1	,			10)
$Tmi1 -0.007** -0.007** \\Tmi10 - [-2.26] \\Tmi10 - [-3.26] \\MTB - [-3.26] \\[13.84] \\Leverage - 0.008*** \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.67] \\[3.68$	$\begin{array}{c} 0.079^{***} \ [3.57] \ 0.006^{***} \ [14.17] \ 0.008^{***} \ [2.57] \ 0.008^{***} \end{array}$	-0.006*		(Q)	(9)	(7)	(8)	(6)
Tmi10   [-2.26] Tmi10   0.006***   [13.84]   Leverage   0.008***   [3.67]   [3.67]	$\begin{array}{c} 0.079^{***} \\ [3.57] \\ 0.006^{***} \\ [14.17] \\ 0.008^{***} \end{array}$		-0.009***		-0.007***	$-0.019^{***}$		-0.018***
Tmi10 MTB 0.006*** [13.84] Leverage 0.008*** [3.67]	$\begin{array}{c} 0.079^{***} \\ [3.57] \\ 0.006^{***} \\ [14.17] \\ 0.008^{***} \end{array}$	-1.34	[-3.40]		[-2.90]	[-3.62]		[-3.52]
$MTB 0.006^{***} \ [13.84] \ Leverage 0.008^{***} \ [3.67]$	$\begin{array}{c} [3.57] \\ 0.006^{***} \\ [14.17] \\ 0.008^{***} \end{array}$	$0.075^{***}$		$0.081^{***}$	$0.076^{***}$	,	0.064	0.049
$ \begin{array}{ccc} MTB & 0.006^{***} & 0 \\ & & [13.84] \\ Leverage & 0.008^{***} \\ & & [3.67] \end{array} $	$\begin{array}{c} 0.006^{***} \\ [14.17] \\ 0.008^{***} \end{array}$	[3.37]		[4.84]	[4.57]		[1.24]	[0.95]
[13.84] Leverage 0.008*** [3.67]	$\begin{bmatrix} 14.17 \\ 0.008^{***} \end{bmatrix}$	$0.006^{***}$	$-0.001^{**}$	$-0.001^{**}$	-0.000	$0.009^{***}$	$0.008^{***}$	$0.009^{***}$
Leverage 0.008*** $[3.67]$	0.008***	[13.98]	[-2.00]	[-2.08]	[-1.59]	[13.14]	[13.08]	[13.13]
[3 67]	1	$0.008^{***}$	-0.028***	-0.028***	-0.028***	$0.019^{***}$	$0.019^{***}$	$0.018^{***}$
- > >	3.65	[3.60]	[-21.31]	[-21.36]	[-21.44]	[5.83]	[6.02]	[5.80]
Cash 0.024***	$0.024^{***}$	$0.024^{***}$	$0.014^{***}$	$0.014^{***}$	$0.014^{***}$	$0.023^{***}$	$0.023^{***}$	$0.023^{***}$
[11.86]	[12.00]	[11.85]	[9.96]	[10.05]	[9.91]	[6.55]	[6.83]	[6.55]
Size -0.008*** -	-0.008***	-0.008***	-0.008***	-0.009***	-0.008***	$-0.010^{***}$	$-0.011^{***}$	$-0.010^{***}$
[-33.04]	[-34.90]	[-32.77]	[-46.53]	[-50.77]	[-46.20]	[-21.29]	[-25.26]	[-21.17]
$Tangibility 0.017^{***}$	$0.017^{***}$	$0.017^{***}$	$0.011^{***}$	$0.011^{***}$	$0.011^{***}$	$0.020^{***}$	$0.021^{***}$	$0.020^{***}$
[5.93]	[6.03]	[5.95]	[4.55]	[4.64]	[4.53]	[3.80]	[3.99]	[3.83]
Age -0.003*** -	-0.003***	-0.003***	-0.005***	-0.005***	-0.005***	$-0.002^{*}$	$-0.002^{*}$	$-0.002^{*}$
[-6.13]	[-6.21]	[-6.18]	[-13.44]	[-13.38]	[-13.53]	[-1.72]	[-1.89]	[-1.74]
Constant 0.119***	$0.119^{***}$	$0.118^{***}$	$0.130^{***}$	$0.129^{***}$	$0.129^{***}$	$0.146^{***}$	$0.148^{***}$	$0.145^{***}$
[25.41]	[25.44]	[25.15]	[36.75]	[36.68]	[36.40]	[14.31]	[14.64]	[14.23]
Observations 80,031	80,031	80,031	45,018	45,018	45,018	35,013	35,013	35,013
Adj. R-squared 0.160	0.160	0.160	0.246	0.246	0.246	0.184	0.184	0.184
Industry fixed effects Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$	$\mathbf{Yes}$	Yes	$\mathbf{Yes}$	$\mathbf{Yes}$
Year fixed effects Yes	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$

investment estimated by the panel regressions. This panel presents the panel regression results of the	nent on the institutional ownership. Our sample consists of 80,031 firm years for the period 1995–2015. The	re the firm inefficient investment proxy variables: $Inef2$ , $Und2$ , and $Ovr2$ estimated by the panel regressions.	iables of interest are the most motivated institutional investor ownership $(Tmi1)$ and the least motivated	ownership $(Tmi10)$ . Detailed definitions of all variables are described in Appendix A. Fama–French 48	d effects are controlled for in all regressions. The standard errors are clustered by firm. t-values are reported	and $*$ denote statistical significance at the 1%, 5%, and 10% levels, respectively.
Panel B. Inefficient investment estimation	firm inefficient investment on the institution	dependent variables are the firm inefficient	The independent variables of interest are	institutional investor ownership $(Tmi10)$ .	industry and year fixed effects are controll	in brackets. * * *, **, and * denote statist

		Inef2			Und2			Ovr2	
Variables	(1)	(2)	(3)	(4)	(2)	(9)	(7)	(8)	(6)
Tmi1	$-0.017^{***}$		$-0.016^{**}$	$-0.007^{***}$		$-0.006^{**}$	$-0.022^{***}$		$-0.021^{***}$
Tmi10		$0.095^{***}$	$0.082^{***}$	[ ]	$0.083^{***}$	0.079*** 14 40		0.060	0.042 0.042
MTB	$0.005^{***}$	$0.005^{***}$	0.005***	-0.000	-0.000 -0.000	[4.40] -0.000	0.008***	0.008***	$0.008^{***}$
Leverage	$[12.97] 0.004^{*}$	$[12.83] \\ 0.004^{*}$	$[13.14] 0.003^{*}$	[-0.44]-0.017***	[-0.40]-0.017***	[-0.01]	$[12.67] 0.010^{***}$	$[12.46] 0.011^{***}$	$[12.65] 0.010^{***}$
Cash	$[1.85] 0.021^{***}$	$[1.95] 0.021^{***}$	$[1.77] 0.021^{***}$	[-12.22] $0.011^{***}$	[-12.30] $0.011^{***}$	[-12.36] $0.011^{***}$	$[3.18] 0.023^{***}$	$[3.43]$ $0.024^{***}$	$[3.16] 0.023^{***}$
Size	[10.77]-0.009***	$[11.10] -0.009^{***}$	[10.76]-0.009***	[7.70]	[7.76] -0.009***	[7.64]-0.009***	[6.88] -0.010***	[7.21]-0.011***	[6.88] -0.010***
Tangibility	$[-37.91] \\ 0.013^{***}$	$\begin{bmatrix} -42.10 \\ 0.013^{***} \end{bmatrix}$	[-37.56] $0.013^{***}$	$[-50.77]$ $0.011^{***}$	$[-55.04]$ $0.011^{***}$	[-50.47] 0.011***	[-20.22] $0.016^{***}$	$[-24.52]$ $0.018^{***}$	$[-20.10]$ $0.017^{***}$
	[4.66]	[4.92] 0.001***	[4.67]	[4.46] 0.004***	[4.52] 0.004***	[4.43] 0.004***	[3.17] 0.000***	[3.39]	[3.19] 0.00e***
Age	[-2.54]	-0.001 [-2.65]	[-2.59]	[10.70]	[10.67]	[10.63]	-0.000 [-8.27]	-0.009 [-8.52]	-0.000 [-8.28]
Constant	$0.127^{***}$ [28.94]	$0.128^{***}$ $[29.32]$	$0.126^{***}$ $[28.63]$	$0.135^{***}$ [38.14]	$0.134^{***}$ $[38.01]$	$0.133^{***}$ $[37.78]$	$0.141^{***}$ $[14.56]$	$0.144^{***}$ $[14.99]$	$0.141^{***}$ $[14.48]$
Observations	80.031	80.031	80.031	45.018	45.018	45.018	35.013	35.013	35.013
Adj. R-squared	0.165	0.164	0.165	0.238	0.238	0.238	0.192	0.192	0.192
Industry fixed effects	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$
Year fixed effects	Yes	${ m Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	${ m Yes}$	$\mathbf{Yes}$	${ m Yes}$	${ m Yes}$

	Š
	2
5	
	Ð.
	Ξ
	5
-	Ξ
	Q
	Б Ц
	5
	g.
	<b>-</b>
5	5
-	Ë
	Ξ.
	Ĕ.
	3
	2
-	Ę
	S
F	
-	÷
	E.
	B
	Ħ
-	St
	ē
	2
	Ξ
	Ξ
	Ð
•	5
ç	Ĕ.
•	
	ă
•	Ш
•	d In
•	nd in
•	and me
•	p and me
•	up and me
• • •	ship and me
• • •	rship and me
• •	nership and ine
• • •	vnership and ine
• • •	ownership and ine
• -	ownership and ine
• • • •	al ownership and ine
• -	nal ownership and me
• -	onal ownership and ine
• • • • • •	tional ownership and ine
• - • -	utional ownership and ine
• • • • • • •	itutional ownership and ine
• • • • • • • • • • • • • • • • • • • •	stitutional ownership and ine
• • •	nstitutional ownership and inc
• F • • F	institutional ownership and inc
• • • • • • • •	d institutional ownership and inc
• • • • • • • • •	ted institutional ownership and inc
• • • • • • • • • • • • • • • • • • • •	ated institutional ownership and inc
• • •	vated institutional ownership and inc
· · · · · · · · · · · · · · · · · · ·	tivated institutional ownership and inc
· · · · · · · · · · · · · · · · · · ·	lotivated institutional ownership and inc
	Motivated institutional ownership and inc
	Motivated institutional ownership and inc
	5: Motivated institutional ownership and inc
	5 5: Motivated institutional ownership and inc
	le 5: Motivated institutional ownership and inc
	where b: Motivated institutional ownership and inc
	lable 5: Motivated institutional ownership and inc

squares (2SLS) regression results of the firm inefficient investment on the institutional ownership. The dependent variable in the first stage regression is the most motivated institutional investor ownership (Tmi1). The instrumental variables used in the first whether the firm drops out of the Russell 2000 index due to the market value decrease, and NTR2 indicating whether the firm gets included in the Russell 2000 index due to the market value increase. Our sample in the first step regressions consists of 92, 546 firm inefficient investment proxy variables: Inef1, Und1, and Ovr1 estimated by the historical panel regressions. In columns 4–6 of Panel A. The institutional ownership level and the inefficient investment level. This panel presents the two stage least stage regressions are the dummy variables: R1TR2 indicating whether the firm switches from the Russell 1000 index into the Russell 2000 index, R2TR1 indicating whether the firm switches from the Russell 2000 index into the Russell 1000 index, R2TN indicating years with data from the Centre for Research in Security Prices, Compustat, and Thomson Financial CDA/Spectrum Institutional (13F) database during the period 1995–2015. In columns 1–3 of the second stage regressions, the dependent variables are the firm the second stage regressions, the dependent variables are the firm inefficient investment proxy variables: Inef2, Und2, and Ovr2estimated by the panel regressions. The independent variable of interest in the second stage regressions is IVTmi1, the predicted Tmi1 by the first stage regressions. Detailed definitions of all variables are described in Appendix A. Fama-French 48 industry and year fixed effects are controlled for in all regressions. The standard errors are clustered by firm. t-values are reported in brackets. \* \* \*, \*\*, and \* denote statistical significance at the 1%, 5%, and 10% levels, respectively.

	First stage			Second	l stage		
		(1)	(2)	(3)	(4)	(5)	(9)
Variables	Tmi1	Inef1	Und1	Ovr1	Inef2	Und2	Ovr2
IVTmi1		$-0.104^{***}$	-0.057***	$-0.151^{***}$	$-0.112^{***}$	-0.063***	$-0.148^{***}$
		[-3.85]	[-3.12]	[-2.95]	[-4.28]	[-3.45]	[-2.95]
R1TR2	$-0.059^{***}$						
	[-14.93]						
R2TR1	$0.041^{***}$						
	[8.67]						
R2TN	$-0.053^{***}$						
	[-33.55]						
NTR2	$-0.033^{***}$						
					5		

	Ι	continued	from prev	ious page			
	First stage			Second	l stage		
		(1)	(2)	(3)	(4)	(5)	(9)
Variables	Tmi1	Inef1	Und1	Ovr1	Inef 2	Und2	Ovr2
	[-17.30]						
MTB	$0.018^{***}$	$0.008^{***}$	0.000	$0.011^{***}$	$0.007^{***}$	$0.001^{**}$	$0.010^{***}$
	[33.16]	[11.24]	[0.91]	[9.28]	[10.67]	[2.19]	[8.96]
Leverage	$-0.025^{***}$	$0.005^{**}$	-0.029***	$0.015^{***}$	0.001	$-0.018^{***}$	$0.007^{**}$
	[-9.21]	[2.41]	[-20.95]	[4.54]	[0.60]	[-12.54]	[2.08]
Cash	$-0.035^{***}$	$0.020^{***}$	$0.012^{***}$	$0.018^{***}$	$0.017^{***}$	$0.008^{***}$	$0.019^{***}$
	[-12.82]	[8.88]	[7.72]	[4.45]	[7.90]	[5.45]	[4.79]
Size	$0.039^{***}$	$-0.004^{***}$	-0.007***	-0.005***	-0.005***	-0.007***	-0.005**
	[56.85]	[-3.92]	[-9.04]	[-2.63]	[-4.99]	[-9.66]	[-2.44]
Tangibility	$-0.040^{***}$	$0.013^{***}$	$0.009^{***}$	$0.015^{***}$	$0.009^{***}$	$0.009^{***}$	$0.012^{**}$
	[-6.88]	[4.16]	[3.58]	[2.61]	[3.00]	[3.37]	[2.05]
Age	0.001	-0.003***	-0.005***	$-0.002^{*}$	$-0.001^{**}$	$0.004^{***}$	-0.008***
	[1.22]	[-5.80]	[-12.99]	[-1.65]	[-2.23]	[11.05]	[-8.23]
Constant	$-0.115^{***}$	$0.108^{***}$	$0.124^{***}$	$0.131^{***}$	$0.116^{***}$	$0.128^{***}$	$0.128^{***}$
	[-13.15]	[18.63]	[29.56]	[11.00]	[21.24]	[30.43]	[11.17]
Observations	92,546	80,031	45,018	35,013	80,031	45,018	35,013
Adj. R-squared	0.379	0.160	0.246	0.184	0.164	0.238	0.192
Industry fixed effects	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$
Year fixed effects	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$

whether the firm switches from the Russell 1000 index into the Russell 2000 index, R2TR1 indicating whether the firm switches due to the market value decrease, and NTR2 indicating whether the firm gets included in the Russell 2000 index due to the market The standard errors are clustered by firm. t-values are reported in brackets. \* \* \*, \*\*, and \* denote statistical significance at the Panel B. The change in institutional ownership and the change in inefficient investment. This panel presents the two from year t-1 to year t. The instrumental variables used in the first stage regressions are the dummy variables: R1TR2 indicating from the Russell 2000 index into the Russell 1000 index, R2TN indicating whether the firm drops out of the Russell 2000 index value increase. Our sample in the first step regressions consists of 83, 778 firm years with data from the Centre for Research in In columns 1–3 of the second stage regressions, the dependent variables are the changes in the firm inefficient investment proxy variables from year t to year t + 1:  $\Delta Inef1$ ,  $\Delta Und1$ , and  $\Delta Ovr1$ . The level of these variables are estimated by the historical panel regressions. In columns 4–6 of the second stage regressions, the dependent variables are the changes in the firm inefficient by the panel regressions. The independent variable of interest in the second stage regressions is the predicted  $\Delta Tmi1$  by the first stage regressions. All the other control variables are the changes of themselves from from year t-1 to year t. Detailed definitions investment proxy variables from year t to year t + 1:  $\Delta Inef2$ ,  $\Delta Und2$ , and  $\Delta Ovr2$ . The level of these variables are estimated of all variables are described in Appendix A. Fama–French 48 industry and year fixed effects are controlled for in all regressions. stage least squares (2SLS) regression results of the change in firm inefficient investment on the change in institutional ownership. The dependent variable in the first stage regression is the change in the most motivated institutional investor ownership (Tmi1)Security Prices, Compustat, and Thomson Financial CDA/Spectrum Institutional (13F) database during the period 1995–2015. 1%, 5%, and 10% levels, respectively.

	First stage			Second	l stage		
		(1)	(2)	(3)	(4)	(5)	(9)
Variables	$\Delta Tmi1$	$\Delta Inef1$	$\Delta Und1$	$\Delta Ovr1$	$\Delta Inef2$	$\Delta Und2$	$\Delta Ovr2$
$IV\Delta Tmi1$		-0.878***	$-0.346^{**}$	-1.587***	$-0.810^{***}$	-0.260	$-1.609^{***}$
		[-3.38]	[-2.17]	[-3.57]	[-3.25]	[-1.62]	[-3.71]
R1TR2	-0.007**						
	[-2.43]						
R2TR1	$0.013^{***}$						
	[3.64]						
R2TN	$0.002^{**}$						
					ŭ	intinued or	n next page

	I	continued	from prev	rious page			
	First stage			Second	l stage		
		(1)	(2)	(3)	(4)	(5)	(9)
Variables	$\Delta Tmi1$	$\Delta Inef1$	$\Delta Und1$	$\Delta Ovr1$	$\Delta Inef2$	$\Delta Und2$	$\Delta Ovr2$
	[2.42]						
NTR2	$0.002^{*}$						
	[1.77]						
$\Delta MTB$	$0.005^{***}$	$0.009^{***}$	$0.002^{**}$	$0.016^{***}$	$0.008^{***}$	0.001	$0.016^{***}$
	[21.19]	[5.95]	[2.36]	[6.50]	[5.77]	[1.40]	[6.81]
$\Delta Leverage$	$-0.007^{***}$	-0.063***	$-0.042^{***}$	-0.088***	-0.060***	-0.033***	-0.094***
	[-4.96]	[-12.56]	[-15.61]	[-10.18]	[-12.91]	[-11.83]	[-11.23]
$\Delta Cash$	$0.010^{***}$	$0.014^{***}$	$0.034^{***}$	0.001	$0.013^{***}$	$0.027^{***}$	0.007
	[9.31]	[3.57]	[14.95]	[0.23]	[3.52]	[11.84]	[1.14]
$\Delta Size$	-0.000	-0.003***	-0.007***	$0.003^{**}$	-0.003***	-0.008***	$0.004^{***}$
	[-0.67]	[-6.36]	[-24.77]	[2.56]	[-7.55]	[-28.41]	[3.14]
$\Delta Tangibility$	$-0.020^{***}$	$-0.023^{*}$	$0.120^{***}$	-0.086***	-0.010	$0.129^{***}$	-0.088***
	[-4.76]	[-1.92]	[13.55]	[-4.21]	[-0.84]	[14.07]	[-4.39]
$\Delta Age$	$0.007^{***}$	0.003	-0.009***	$-0.034^{***}$	$0.006^{*}$	$0.015^{***}$	$-0.052^{***}$
	[3.38]	[0.99]	[-3.62]	[-4.22]	[1.70]	[5.83]	[-6.75]
Constant	0.001	$0.006^{**}$	-0.008***	$-0.019^{***}$	$0.008^{***}$	$-0.012^{***}$	$-0.016^{**}$
	[0.44]	[2.50]	[-3.81]	[-2.74]	[3.46]	[-5.44]	[-2.37]
Observations	83,778	73,466	40,880	32,586	73,466	40,880	32,586
Adj. R-squared	0.018	0.018	0.086	0.035	0.020	0.089	0.037
Industry fixed effects	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$
Year fixed effects	$\mathbf{Y}_{\mathbf{es}}$						

Table 6: Motivated institutional ownership by institutional investor types

 $Tmi1\_Ind$  and motivated grey institutional ownership  $Tmi1\_Grey$ . In columns (1) and (4), the dependent variables in the first variables: R1TR2 indicating whether the firm switches from the Russell 1000 index into the Russell 2000 index, R2TR1 indicating whether the firm switches from the Russell 2000 index into the Russell 1000 index, R2TN indicating whether the firm drops out of the Russell 2000 index due to the market value decrease, and NTR2 indicating whether the firm gets included in the Russell 2000 index due to the market value increase. Our sample in the first step regressions consists of 92, 546 firm years with data from the investment proxy variables: Inef1 estimated by the historical panel regressions and Inef2 estimated by the panel regressions. The independent variables of interest in the second stage regressions are IVTmi1\_Ind and IVTmi1\_Grey, the predicted ownership by the first stage regressions. Detailed definitions of all variables are described in Appendix A. Fama–French 48 industry and year This panel presents the two stage least squares (2SLS) regression results of the firm inefficient investment on the institutional ownership by types. We divide total motivated institutional ownership Tmi1 into motivated independent institutional ownership stage regressions are Tmi1-Ind and Tmi1-Grey. The instrumental variables used in the first stage regressions are the dummy Centre for Research in Security Prices, Compustat, and Thomson Financial CDA/Spectrum Institutional (13F) database during the period 1995–2015. In columns (2) and (4), the dependent variables in the second stage regressions are the firm inefficient fixed effects are controlled for in all regressions. The standard errors are clustered by firm. t-values are reported in brackets. \*\*\*, grey investors. Panel A. The institutional ownership level and the inefficient investment level: Independent vs. \*\*, and \* denote statistical significance at the 1%, 5%, and 10% levels, respectively.

					i	
	First stage	Second	l stage	First stage	Second	stage
	(1)	(2)	(3)	(4)	(5)	(9)
Variables	Tmi1Ind	Inef 1	Inef 2	$Tmi1\_Grey$	Inef1	Inef 2
$IVTmi1\_Ind$		$-0.140^{***}$	$-0.112^{***}$			
		[-3.95]	[-4.28]			
$IVTmi1\_Grey$					-0.281***	-0.309***
					[-3.12]	[-3.58]
R1TR2	-0.038***			$-0.021^{***}$		
	[-12.40]			[-10.41]		
R2TR1	-0.038***			$-0.016^{***}$		
	[-28.83]			[-23.13]		
				Ŭ	ontinued on	next page

	- con	tinued fror	n previous	s page		
	First stage	Second	l stage	First stage	Second	l stage
	(1)	(2)	(3)	(4)	(5)	(9)
Variables	Tmi1Ind	Inef1	Inef2	$Tmi1\_Grey$	Inef1	Inef2
R2TN	$0.046^{***}$			-0.005***		
NTR2	[11.64]-0.021***			[-2.88] -0.012***		
MTB	[-13.09] $0.013^{***}$	$0.008^{***}$	0.007***	[-14.99] $0.005^{***}$	0.007***	0.007***
Leveraae	[32.03]-0.017***	$[11.50] 0.005^{**}$	[10.67] 0.001	[23.64]-0.007***	$[11.13] 0.006^{***}$	[10.53] 0.002
Cash	[-8.16] -0.019***	[2.51]0.021***	[0.60] 0.017***	[-6.78] -0.016***	[2.62] 0.019***	[0.82] 0.016***
Size	[-8.78] [-878]	[9.93] 	[7.90] 	[-14.88] 0.019***	[7.44] 006***	[6.47] 
Tan aibilitu	[57.01] 	[-4.58] 0.013***	[-4.99] 0.000***	[39.27] [0.019***	[-4.60]	[-5.63]
Anne z		[4.31] [0.004***	[3.00] -0.001**	[-5.09] 0.003***	[4.42] -0.003***	[3.24] -0.000
Con stant	[-2.54] 	[-6.58] 0.110***	[-2.23]	[6.71] 	[-3.90]	[-0.60]
	[-9.31]	[20.21]	[21.24]	[-13.61]	[17.00]	[19.26]
Observations	92,546	80,031	80,031	92,546	80,031	80,031
Adj. R-squared	0.328	0.160	0.164	0.241	0.160	0.164
Industry fixed effects	${ m Yes}_{{ m Ves}}$	${ m Yes}_{{ m V}_{22}}$	${ m Yes}_{ m voc}$	${ m Yes}_{ m Voc}$		${ m Yes}_{{ m V}_{22}}$
Year nxed enects	res	Yes	res	Yes		Yes

the change in institutional ownership by types. We divide total motivated institutional ownership Tmi1 into motivated independent  $\Delta Inef1$  estimated by the historical panel regressions and  $\Delta Inef2$  estimated by the panel regressions. The independent variables of interest in the second stage regressions are  $\Delta IVTmi1$  and  $\Delta IVTmi1$ . Grey, the predicted ownership changes by the first stage regressions. All the other control variables are the changes of themselves from from year t-1 to year t. Detailed definitions The standard errors are clustered by firm. t-values are reported in brackets. \* \* \*, \*\*, and \* denote statistical significance at the investors. This panel presents the two stage least squares (2SLS) regression results of the change in firm inefficient investment on institutional ownership Tmi1-Ind and motivated grey institutional ownership Tmi1-Grey. In columns (1) and (4), the dependent variables in the first stage regressions are the changes in Tmi1 and Tmi1. Grey, from year t-1 to year t. The instrumental variables used in the first stage regressions are the dummy variables: R1TR2 indicating whether the firm switches from the Russell 1000 index into the Russell 2000 index, R2TR1 indicating whether the firm switches from the Russell 2000 index into the Russell 1000 index, R2TN indicating whether the firm drops out of the Russell 2000 index due to the market value decrease, and NTR2indicating whether the firm gets included in the Russell 2000 index due to the market value increase. Our sample in the first step regressions consists of 84, 731 firm years with data from the Centre for Research in Security Prices, Compustat, and Thomson Financial CDA/Spectrum Institutional (13F) database during the period 1995–2015. In columns (2) and (4), the dependent variables in the second stage regressions are the changes in the firm inefficient investment proxy variables from year t to year t+1. of all variables are described in Appendix A. Fama–French 48 industry and year fixed effects are controlled for in all regressions. Panel B. The change in institutional ownership and the change in inefficient investment: Independent vs. grey 1%, 5%, and 10% levels, respectively.

	First stage	Second	l stage	First stage	Second	l stage
	(1)	(2)	(3)	(4)	(5)	(9)
Variables	$\Delta Tmi1\_Ind$	$\Delta Inef1$	$\Delta Inef2$	$\Delta Tmi1\_Grey$	$\Delta Inef1$	$\Delta Inef2$
$IV\Delta Tmi1\_Ind$		-1.068***	-0.972***			
		[-3.48]	[-3.31]			
$IV\Delta Tmi1\_Grey$			-		$-1.647^{**}$	$-1.592^{**}$
					[-2.18]	[-2.25]
R1TR2	-0.003			-0.004***	,	1
	[-1.54]			[-2.96]		
R2TR1	0.001			0.001		
				Ŭ	ontinued on	next page

	- COI	ntinued fro	m previou	s page		
	First stage	Second	l stage	First stage	Second	l stage
	(1)	(2)	(3)	(4)	(2)	(9)
Variables	$\Delta Tmi1$ Ind	$\Delta Inef1$	$\Delta Inef2$	$\Delta Tmi1\_Grey$	$\Delta Inef1$	$\Delta Inef 2$
	[20.0]			[1.35]		
R2TN	$0.012^{***}$			0.000		
	[4.03]			[0.05]		
NTR2	0.001			0.000		
	[1.34]			[0.52]		
$\Delta MTB$	$0.004^{***}$	$0.009^{***}$	$0.008^{***}$	$0.001^{***}$	$0.006^{***}$	$0.005^{***}$
	[19.51]	[6.16]	[5.92]	[9.38]	[6.56]	[6.56]
$\Delta Leverage$	-0.006***	$-0.064^{***}$	$-0.061^{***}$	-0.001	-0.059***	-0.057***
	[-5.33]	[-12.57]	[-12.88]	[-1.47]	[-12.38]	[-12.83]
$\Delta Cash$	$0.009^{***}$	$0.015^{***}$	$0.014^{***}$	0.001	$0.006^{**}$	$0.006^{**}$
	[9.71]	[3.68]	[3.59]	[1.45]	[2.11]	[2.17]
$\Delta Size$	0.000	-0.003***	-0.003***	-0.000*	-0.003***	-0.004***
	[0.55]	[-5.54]	[-6.76]	[-1.92]	[-6.27]	[-7.37]
$\Delta T angibility$	$-0.017^{***}$	$-0.024^{**}$	-0.011	-0.002	-0.010	0.003
	[-5.15]	[-2.01]	[-0.92]	[-1.24]	[-0.86]	[0.25]
$\Delta Age$	$0.003^{*}$	0.000	0.003	$0.004^{***}$	0.004	0.006
	[1.92]	[0.16]	[1.00]	[3.58]	[0.88]	[1.55]
Constant	$0.003^{**}$	$0.009^{***}$	$0.010^{***}$	-0.003***	0.000	0.002
	[2.13]	[3.42]	[4.22]	[-3.45]	[0.10]	[0.80]
		001 01				
<b>U</b> DServationS	$\delta 4, \ell 31$	13,400	13,400	84,731	(3,400)	13,400
Adj. R-squared	0.016	0.018	0.020	0.011	0.018	0.020
Industry fixed effects	${ m Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$	${ m Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$
Year fixed effects	$\mathrm{Yes}$	$Y_{es}$	$\mathbf{Yes}$	${ m Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$

R2TR1 indicating whether the firm switches from the Russell 2000 index into the Russell 1000 index, R2TN indicating whether in the Russell 2000 index due to the market value increase. Our sample in the first step regressions consists of 92, 546 firm years regressions. The independent variables of interest in the second stage regressions are *IVTmi1\_Tran* and *IVTmi1\_NonTran*, the 48 industry and year fixed effects are controlled for in all regressions. The standard errors are clustered by firm. t-values are tional ownership by types. We divide total motivated institutional ownership Tmi1 into motivated transient institutional ownership  $Tmi1\_Tran$  and motivated non-transient institutional ownership  $Tmi1\_NonTran$ . In columns (1) and (4), the dependent variables in the first stage regressions are  $Tmi1\_Tran$  and  $Tmi1\_NonTran$ . The instrumental variables used in the first stage regressions are the dummy variables: R1TR2 indicating whether the firm switches from the Russell 1000 index into the Russell 2000 index, the firm drops out of the Russell 2000 index due to the market value decrease, and NTR2 indicating whether the firm gets included database during the period 1995–2015. In columns (2) and (4), the dependent variables in the second stage regressions are the firm inefficient investment proxy variables: Inef1 estimated by the historical panel regressions and Inef2 estimated by the panel predicted ownership by the first stage regressions. Detailed definitions of all variables are described in Appendix A. Fama-French vestors. This panel presents the two stage least squares (2SLS) regression results of the firm inefficient investment on the institu-Panel C. The institutional ownership level and the inefficient investment level: Transient vs. non-transient inwith data from the Centre for Research in Security Prices, Compustat, and Thomson Financial CDA/Spectrum Institutional (13F) reported in brackets. \* \* \*, \*\*, and \* denote statistical significance at the 1%, 5%, and 10% levels, respectively.

	First stage	Second	l stage	First stage	Second	l stage
	(1)	(2)	(3)	(4)	(5)	(9)
Variables	$Tmil_Tran$	Inef 1	Inef2	$Tmi1\_NonTran$	Inef 1	Inef 2
$iVTmi1\_Tran$		-0.237***	$-0.245^{***}$			
		[-3.62]	[-3.90]			
$iVTmi1\_NonTran$					$-0.132^{***}$	$-0.143^{***}$
					[-3.50]	[-3.95]
R1TR2	$-0.012^{***}$			$-0.046^{***}$	1	1
	[-8.23]			[-13.30]		
R2TR1	$-0.013^{***}$			$-0.039^{***}$		
	[-16.74]			[-32.21]		
R2TN	$0.037^{***}$			0.005		
				Ŭ	ontinued on	next page

	- 00	ntinued fr	om previo	us page		
	First stage	Second	l stage	First stage	Second	l stage
	(1)	(2)	(3)	(4)	(5)	(9)
Variables	$Tmil_Tran$	Inef1	Inef2	$Tmi1\_NonTran$	Inef1	Inef2
	[15.93]			[1.27]		
NTR2	-0.004***			-0.029***		
	[-3.87]			[-21.05]		
MTB	$0.005^{***}$	$0.007^{***}$	$0.006^{***}$	$0.013^{***}$	$0.008^{***}$	$0.007^{***}$
	[27.78]	[12.16]	[11.29]	[30.39]	[11.22]	[10.63]
Leverage	$-0.002^{**}$	$0.007^{***}$	$0.003^{*}$	-0.022***	$0.005^{**}$	0.001
	[-2.34]	[3.47]	[1.76]	[-10.55]	[2.16]	[0.35]
Cash	$0.002^{**}$	$0.025^{***}$	$0.022^{***}$	-0.037***	$0.019^{***}$	$0.016^{***}$
	[2.47]	[12.28]	[11.41]	[-17.07]	[7.42]	[6.45]
Size	$0.009^{***}$	-0.006***	-0.007***	$0.030^{***}$	-0.004***	-0.005***
	[56.28]	[-9.97]	[-12.36]	[51.10]	[-3.90]	[-4.86]
Tangibility	-0.009***	$0.015^{***}$	$0.011^{***}$	$-0.030^{***}$	$0.013^{***}$	$0.009^{***}$
	[-5.61]	[5.09]	[3.96]	[-6.28]	[4.21]	[3.03]
Age	-0.003***	-0.004***	-0.002***	$0.004^{***}$	-0.003***	-0.001
	[-8.51]	[-7.09]	[-3.67]	[5.17]	[-4.73]	[-1.25]
Constant	-0.023***	$0.115^{***}$	$0.123^{***}$	-0.092***	$0.108^{***}$	$0.116^{***}$
	[-9.72]	[23.02]	[26.49]	[-11.98]	[17.96]	[20.42]
Observations	92,546	80,031	80,031	92,546	80,031	80,031
Adj. R-squared	0.172	0.160	0.164	0.350	0.160	0.164
Industry fixed effects	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Yes}$	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$
Year fixed effects	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$

investment on the change in institutional ownership by types. We divide total motivated institutional ownership Tmi1 into motivated transient institutional ownership  $Tmi1\_Tran$  and motivated non-transient institutional ownership  $Tmi1\_NonTran$ . In whether the firm switches from the Russell 1000 index into the Russell 2000 index, R2TR1 indicating whether the firm switches columns (1) and (4), the dependent variables in the first stage regressions are the changes in Tmi1-Tran and Tmi1-NonTran, from year t-1 to year t. The instrumental variables used in the first stage regressions are the dummy variables: R1TR2 indicating from the Russell 2000 index into the Russell 1000 index, R2TN indicating whether the firm drops out of the Russell 2000 index due to the market value decrease, and NTR2 indicating whether the firm gets included in the Russell 2000 index due to the market value increase. Our sample in the first step regressions consists of 84, 731 firm years with data from the Centre for Research in In columns (2) and (4), the dependent variables in the second stage regressions are the changes in the firm inefficient investment proxy variables from year t to year t+1::  $\Delta Inef1$  estimated by the historical panel regressions and  $\Delta Inef2$  estimated by the panel regressions. The independent variables of interest in the second stage regressions are  $\Delta IVTmi1$ -Tran and  $\Delta IVTmi1$ -NonTran, the predicted ownership changes by the first stage regressions. All the other control variables are the changes of themselves from from year t-1 to year t. Detailed definitions of all variables are described in Appendix A. Fama–French 48 industry and year fixed effects are controlled for in all regressions. The standard errors are clustered by firm. t-values are reported in brackets. \*\*\*, \*\*, transient investors. This panel presents the two stage least squares (2SLS) regression results of the change in firm inefficient Panel D. The change in institutional ownership and the change in inefficient investment: Transient vs. non-Security Prices, Compustat, and Thomson Financial CDA/Spectrum Institutional (13F) database during the period 1995–2015. and \* denote statistical significance at the 1%, 5%, and 10% levels, respectively.

	First stage	Second	l stage	First stage	Second	l stage
	(1)	(2)	(3)	(4)	(5)	(9)
Variables	$\Delta Tmi1Tran$	$\Delta I nef1$	$\Delta Inef2$	$\Delta Tmi1$ _NonTran	$\Delta Inef1$	$\Delta Inef2$
$IV\Delta Tmi1\_Tran$		$-2.140^{***}$	$-1.910^{***}$			
		[-3.92]	[-3.65]			
$IV\Delta Tmi1\_NonTran$		1	1		$-1.421^{***}$	-1.298***
					[-3.57]	[-3.42]
R1TR2	$-0.002^{*}$			$-0.005^{**}$	,	1
	[-1.72]			[-2.45]		
R2TR1	0.000			-0.000		
				C	ontinued on	next page

	- 60	ntinued fr	om previo	us page		
	First stage	Second	l stage	First stage	Second	l stage
	(1)	( <b>2</b> )	(3)	(4)	(5)	(9)
Variables	$\Delta Tmi1Tran$	$\Delta Inef1$	$\Delta Inef2$	$\Delta Tmi1_NonTran$	$\Delta Inef1$	$\Delta Inef2$
	[0.36]			[-0.66]		
R2TN	$0.006^{***}$			$0.009^{***}$		
	[3.45]			[3.33]		
NTR2	$0.002^{**}$			-0.000		
	[2.03]			[-0.26]		
$\Delta MTB$	$0.003^{***}$	$0.010^{***}$	$0.009^{***}$	$0.003^{***}$	$0.008^{***}$	$0.007^{***}$
	[18.15]	[6.37]	[6.04]	[16.64]	[6.86]	[6.64]
$\Delta Leverage$	-0.003***	-0.063***	-0.060***	$-0.005^{***}$	-0.065***	-0.062***
	[-3.56]	[-12.81]	[-13.13]	[-5.51]	[-12.54]	[-12.84]
$\Delta Cash$	$0.006^{***}$	$0.018^{***}$	$0.017^{***}$	$0.004^{***}$	$0.010^{***}$	$0.010^{***}$
	[9.08]	[4.09]	[3.91]	[5.83]	[3.26]	[3.22]
$\Delta Size$	-0.000	-0.003***	-0.003***	0.000*	-0.002***	-0.003***
	[-0.70]	[-6.44]	[-7.61]	[1.67]	[-4.35]	[-5.55]
$\Delta T angibility$	-0.009***	$-0.024^{**}$	-0.010	$-0.011^{***}$	$-0.021^{*}$	-0.008
	[-3.44]	[-2.03]	[-0.89]	[-3.91]	[-1.82]	[-0.71]
$\Delta Age$	0.000	-0.002	0.000	-0.001	-0.004	-0.001
	[0.18]	[-0.87]	[0.08]	[-0.33]	[-1.44]	[-0.45]
Constant	0.001	$0.007^{***}$	$0.009^{***}$	0.001	$0.007^{***}$	$0.009^{***}$
	[0.78]	[2.96]	[3.85]	[1.22]	[2.98]	[3.89]
-						
Observations	84,731	73,466	73,466	84,731	73,466	73,466
Adj. R-squared	0.014	0.018	0.020	0.010	0.018	0.020
Industry fixed effects	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$
Year fixed effects	Yes	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes

# Table 7: How do motivated monitoring institutional investors mitigate firm over-investment?

This table presents the second-stage regressions of firm over-investment on the product of the predicated motivated institutional ownership and firm cash. The first step regression is the same as the one in Panel A of Table 5. In columns 1 and 3, the dependent variables are the firm over-investment proxy variable Ovr1, estimated by the historical panel regressions. In columns 2 and 4, the dependent variables are the firm over-investment proxy variable Ovr2, estimated by the panel regressions. The independent variables of interest in the second stage regressions are the product of the predicted Tmi1 by the first stage regressions and Cash (IVTmi1 \* Cash) and the product of the predicted Tmi1 by the first stage regressions and FCF (IVTmi1 \* FCF). FCF1 (FCF2) are estimated by the historical panel regressions (the panel regressions). Detailed definitions of all variables are described in Appendix A. Fama–French 48 industry and year fixed effects are controlled for in all regressions. The standard errors are clustered by firm. t-values are reported in brackets. \* \* \*, \*\*, and \* denote statistical significance at the 1%, 5%, and 10% levels, respectively.

	(1)	(2)	(3)	(4)
Variables	Ovr1	Ovr2	Ovr1	Ovr2
IVTmi1	-0.106**	-0.106**	-0.128*	-0.132*
	[-2.02]	[-2.07]	[-1.86]	[-1.82]
IVTmi1*Cash	-0.130***	-0.120***		
	[-3.80]	[-3.58]		
IVTmi1 * FCF1			-0.264**	
			[-2.06]	
FCF1			0.128***	
			[5.09]	0.000**
IVTmi1 * FCF2				-0.302**
ECE				[-2.28]
$F \cup F Z$				[4.66]
Cash	0 025***	0.025***	0.013**	[4.00] 0.011*
Cush	[5, 67]	[5.00]	[2 00]	$\begin{bmatrix} 1 & 73 \end{bmatrix}$
MTB	0.011***	0.011***	0.008***	0.008***
	[9 40]	[9 07]	[4 85]	[4 57]
Leveraae	0.016***	0.007**	0.013***	0.007
2000, ago	[4.62]	[2.16]	[3.22]	[1.61]
Size	-0.006***	-0.006***	-0.005**	-0.005*
	[-3.05]	[-2.84]	[-1.98]	[-1.82]
Tangibility	0.015***	0.012**	0.022***	0.016**
	[2.66]	[2.09]	[3.16]	[2.23]
Age	-0.002**	-0.009***	-0.003**	-0.008***
	[-2.09]	[-8.69]	[-2.14]	[-6.32]
Constant	$0.132^{***}$	$0.128^{***}$	$0.131^{***}$	$0.131^{***}$
	[11.14]	[11.31]	[8.14]	[7.97]
Observations	35,013	35,013	19,333	17,529
Adj. R-squared	0.185	0.192	0.134	0.144
Industry fixed effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes

# Table 8: How do motivated monitoring institutional investors mitigate firm under-investment?

This table presents the second-stage regressions of firm under-investment on the product of the predicated motivated institutional ownership and the variables proxy for firm managers' career concern. The first step regression is the same as the one in Panel A of Table 5. In columns 1 and 3, the dependent variables are the firm under-investment proxy variable Und1, estimated by the historical panel regressions. In columns 2 and 4, the dependent variables are the firm under-investment proxy variable Und1, estimated by the historical panel regressions. In columns 2 and 4, the dependent variables are the firm under-investment proxy variable Und2, estimated by the panel regressions. In columns (1) and (2), the independent variable of interest is the product of the predicted Tmi1 by the first stage regressions and Competition (IVTmi1 \* Competition). In columns (3) and (4), the independent variable of interest is the product of the predicted Tmi1 by the first stage regressions and GIndex (IVTmi1 \* GIndex). Detailed definitions of all variables are described in Appendix A. Fama–French 48 industry and year fixed effects are controlled for in all regressions. The standard errors are clustered by firm. t-values are reported in brackets. \* \* \*, \*\*, and \* denote statistical significance at the 1%, 5%, and 10% levels, respectively.

	(1)	(2)	(3)	(4)
Variables	Und1	$(\mathbf{Z})$ Und2	Und1	Und2
IUTmi1	0.020	0.022	0.116**	0.110**
1 / 1 ////1	-0.029 [1.15]	-0.033	[2.00]	-0.119
$WT = i1 \cdot C = i + i + i = i$	[-1.10]	[-1.20]	[-2.00]	[-1.99]
<i>IVI mil</i> * <i>Competition</i>	-0.059	-0.062		
<i>a</i>	[-2.29]	[-2.34]		
Competition	0.035***	0.034***		
	[4.94]	[4.65]	0.000k	0.0004
IVTmi1 * GIndex			0.008*	0.008*
			[1.65]	[1.80]
GIndex			-0.002**	-0.002**
			[-2.01]	[-2.13]
MTB	-0.001***	-0.001**	0.000	0.001
	[-2.83]	[-2.17]	[0.23]	[0.64]
Leverage	-0.032***	-0.027***	-0.029***	-0.015***
	[-23.57]	[-18.32]	[-7.51]	[-3.68]
Cash	0.016***	0.013***	0.007	0.006
	[10.03]	[7.70]	[1.40]	[1.20]
Size	-0.004***	-0.005***	-0.007***	-0.008***
	[-6.15]	[-6.57]	[-3.87]	[-4.21]
Tanaibility	0.012***	0.011***	0.021***	0.023***
	[4.06]	[3,79]	[2.66]	[2.87]
Aae	-0.001**	0.006***	-0.007***	0.003***
	[-2.52]	[13.55]	[-7.55]	[3.46]
Constant	0.047***	0.051***	0 136***	0 141***
Constant	[6,39]	[09 6]	[10.97]	[11.53]
	[0.00]	[0.50]	[10.01]	[11.00]
Observations	47.445	47.445	3.082	3.082
R-squared	0.198	0.181	0.251	0.265
Industry fixed effects	Yes	Yes	Yes	Yes
Year fixed effects	Yes	Yes	Yes	Yes
FOR HACK CHOOD	100	100	100	100

•	up
	ownerst
	tor
•	Inves
-	ਸ਼
•	ltion
•	citu
	Inst
•	rıng
	Ito
	mon
	ated
•	notiva
c	ot
	lres
	meası
•	lative measi
•	Alternative measu
	9: Alternative measu
	ble 9: Alternative measu

indicating whether the firm switches from the Russell 2000 index into the Russell 1000 index, R2TN indicating whether the firm drops out of the Russell 2000 index due to the market value decrease, and NTR2 indicating whether the firm gets included in the Russell 2000 index due to the market value increase. Our sample in the first step regressions consists of 92, 546 firm years with data the panel regressions. The independent variable of interest in the second stage regressions is IVPmi1, the predicted Pmi1 by the Panel A. The ratio of the most motivated monitoring institutional investor number to the total institutional investor number. This panel presents the two stage least squares (2SLS) regression results of the firm inefficient investment on the institutional ownership. The dependent variable in the first stage regression is the ratio of the most motivated institutional investor number to the total institutional investor number (Pmi1). The instrumental variables used in the first stage regressions are from the Centre for Research in Security Prices, Compustat, and Thomson Financial CDA/Spectrum Institutional (13F) database during the period 1995–2015. In columns 1–3 of the second stage regressions, the dependent variables are the firm inefficient investment proxy variables: Inef1, Und1, and Ovr1 estimated by the historical panel regressions. In columns 4–6 of the second stage regressions, the dependent variables are the firm inefficient investment proxy variables: Inef2, Und2, and Ovr2 estimated by first stage regressions. Detailed definitions of all variables are described in Appendix A. Fama-French 48 industry and year fixed the dummy variables: R1TR2 indicating whether the firm switches from the Russell 1000 index into the Russell 2000 index, R2TR1effects are controlled for in all regressions. The standard errors are clustered by firm. t-values are reported in brackets. \* \* \*, \*\*, and \* denote statistical significance at the 1%, 5%, and 10% levels, respectively.

	First stage			Second	l stage		
		(1)	(2)	(3)	(4)	(2)	(9)
Variables	Pmi1	Inef1	Und1	Ovr1	Inef 2	Und2	Ovr2
IVPmi1		-0.265***	$-0.145^{***}$	-0.392***	-0.288***	-0.172***	-0.382***
		[-3.41]	[-2.84]	[-2.63]	[-3.87]	[-3.33]	[-2.62]
R1TR2	$-0.017^{***}$	1	1	1	1	1	1
	[-14.61]						
R2TR1	$-0.017^{***}$						
	[-29.90]						
R2TN	$0.023^{***}$						
	[13.78]						
					O	ontinued on	next page

		continued	from prev	ious page			
	First stage			Second	l stage		
		(1)	(2)	(3)	(4)	(2)	(9)
Variables	Pmi1	Inef1	Und1	Ovr1	Inef2	Und2	Ovr2
NTR2	$-0.011^{***}$						
	[-16.84]						
MTB	$0.007^{***}$	$0.008^{***}$	0.000	$0.011^{***}$	$0.007^{***}$	$0.001^{**}$	$0.010^{***}$
	[25.70]	[10.48]	[0.90]	[8.47]	[10.01]	[2.25]	[8.18]
Leverage	$-0.012^{***}$	$0.005^{**}$	$-0.030^{***}$	$0.015^{***}$	0.001	$-0.018^{***}$	$0.006^{*}$
	[-10.27]	[2.11]	[-20.41]	[4.04]	[0.30]	[-12.40]	[1.74]
Cash	$-0.013^{***}$	$0.020^{***}$	$0.012^{***}$	$0.018^{***}$	$0.017^{***}$	$0.008^{***}$	$0.019^{***}$
	[-10.00]	[8.80]	[7.71]	[4.37]	[7.81]	[5.36]	[4.71]
Size	$0.013^{***}$	-0.005***	-0.007***	-0.006***	-0.006***	-0.007***	-0.006***
	[34.88]	[-4.46]	[-9.62]	[-2.94]	[-5.53]	[-10.09]	[-2.76]
Tangibility	$-0.013^{***}$	$0.013^{***}$	$0.010^{***}$	$0.016^{***}$	$0.009^{***}$	$0.009^{***}$	$0.013^{**}$
	[-5.88]	[4.45]	[3.76]	[2.80]	[3.28]	[3.51]	[2.24]
Age	$0.001^{**}$	-0.003***	-0.005***	-0.002	$-0.001^{**}$	$0.004^{***}$	-0.008***
	[2.15]	[-5.53]	[-12.72]	[-1.46]	[-1.96]	[11.20]	[-7.98]
Constant	-0.024***	$0.114^{***}$	$0.127^{***}$	$0.139^{***}$	$0.122^{***}$	$0.131^{***}$	$0.136^{***}$
	[-3.51]	[22.14]	[33.46]	[12.84]	[25.39]	[34.49]	[13.11]
Observations	92,546	80,031	45,018	35,013	80,031	45,018	35,013
Adj. R-squared	0.328	0.160	0.246	0.184	0.164	0.238	0.192
Industry fixed effects	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$
Year fixed effects	$\mathbf{Yes}$	$\mathbf{Yes}$	Yes	$\mathbf{Y}_{\mathbf{es}}$	Yes	$\mathbf{Yes}$	$\mathbf{Yes}$

Panel B. The natural log of one plus the most motivated institutional investor number (Nmi1): Ln(1+Nmi1). This whether the firm switches from the Russell 1000 index into the Russell 2000 index, R2TR1 indicating whether the firm switches due to the market value decrease, and NTR2 indicating whether the firm gets included in the Russell 2000 index due to the market variables are the firm inefficient investment proxy variables: Inef2, Und2, and Ovr2 estimated by the panel regressions. The independent variable of interest in the second stage regressions is IVLn(1+Nmi1), the predicted Ln(1+Nmi1) by the first stage regressions. Detailed definitions of all variables are described in Appendix A. Fama–French 48 industry and year fixed effects are controlled for in all regressions. The standard errors are clustered by firm. t-values are reported in brackets. \*\*\*, \*\*, and \* denote The dependent variable in the first stage regression is Ln(1 + Nmi1): the natural log of one plus the most motivated institutional investor number (Nmi1). The instrumental variables used in the first stage regressions are the dummy variables: R1TR2 indicating from the Russell 2000 index into the Russell 1000 index, R2TN indicating whether the firm drops out of the Russell 2000 index value increase. Our sample in the first step regressions consists of 92,546 firm years with data from the Centre for Research in In columns 1–3 of the second stage regressions, the dependent variables are the firm inefficient investment proxy variables: Inef1, Und1, and Ovr1 estimated by the historical panel regressions. In columns 4-6 of the second stage regressions, the dependent panel presents the two stage least squares (2SLS) regression results of the firm inefficient investment on the institutional ownership. Security Prices, Compustat, and Thomson Financial CDA/Spectrum Institutional (13F) database during the period 1995–2015. statistical significance at the 1%, 5%, and 10% levels, respectively.

	First stage			Second	l stage		
		(1)	(2)	(3)	(4)	(2)	(9)
Variables	Ln(1+Nmi1)	Inef1	Und1	Ovr1	Inef2	Und2	Ovr2
IVLn(1 + Nmi1)		$-0.011^{***}$	-0.006***	$-0.016^{***}$	$-0.012^{***}$	-0.007***	$-0.015^{***}$
		[-4.11]	[-3.33]	[-3.04]	[-4.49]	[-3.54]	[-3.03]
R1TR2	$-0.070^{**}$						
	[-2.36]						
R2TR1	$-0.551^{***}$						
	[-49.14]						
R2TN	$1.134^{***}$						
	[37.22]						
NTR2	-0.278***						
					C	an Louistan	nort now

	-	continued f	rom previ	ous page			
	First stage			Second	l stage		
		(1)	(2)	(3)	(4)	(5)	(9)
Variables	Ln(1+Nmi1)	Inef1	Und1	Ovr1	Inef2	Und2	Ovr2
	[-20.76]						
MTB	$0.180^{***}$	$0.008^{***}$	0.001	$0.011^{***}$	$0.007^{***}$	$0.001^{**}$	$0.010^{***}$
	[38.09]	[11.27]	[1.16]	[9.33]	[10.68]	[2.31]	[9.01]
Leverage	$-0.266^{***}$	$0.005^{**}$	$-0.030^{***}$	$0.015^{***}$	0.001	$-0.018^{***}$	$0.007^{**}$
	[-12.89]	[2.19]	[-20.81]	[4.41]	[0.39]	[-12.52]	[1.97]
Cash	-0.338***	$0.020^{***}$	$0.012^{***}$	$0.018^{***}$	$0.017^{***}$	$0.008^{***}$	$0.019^{***}$
	[-15.29]	[8.89]	[7.62]	[4.57]	[7.94]	[5.42]	[4.92]
Size	$0.366^{***}$	$-0.004^{***}$	-0.006***	-0.006***	-0.005***	-0.007***	-0.005***
	[61.84]	[-3.97]	[-8.98]	[-2.91]	[-5.13]	[-9.73]	[-2.71]
Tangibility	$-0.346^{***}$	$0.013^{***}$	$0.009^{***}$	$0.016^{***}$	$0.009^{***}$	$0.009^{***}$	$0.012^{**}$
	[-7.53]	[4.31]	[3.63]	[2.78]	[3.16]	[3.46]	[2.21]
Age	$0.041^{***}$	-0.003***	-0.005***	-0.001	-0.001	$0.004^{***}$	-0.008***
	[4.83]	[-5.06]	[-12.29]	[-1.20]	[-1.55]	[11.37]	[-7.60]
Constant	$-1.125^{***}$	$0.107^{***}$	$0.123^{***}$	$0.131^{***}$	$0.115^{***}$	$0.127^{***}$	$0.127^{***}$
	[-14.48]	[18.45]	[29.24]	[11.05]	[21.08]	[30.16]	[11.23]
Observations	92,546	80,031	45,018	35,013	80,031	45,018	35,013
Adj. R-squared	0.513	0.160	0.246	0.184	0.164	0.238	0.192
Industry fixed effects	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$
Year fixed effects	Yes	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Yes}$	Yes	$\mathbf{Yes}$	$\mathbf{Y}_{\mathbf{es}}$	$\mathbf{Y}_{\mathbf{es}}$